ERIA Discussion Paper Series

Japanese Small and Medium-Sized Enterprises' Export Decisions: The Role of Overseas Market Information*

Tomohiko INUI[†]

Preparatory Office for the Faculty of International Social Studies, Gakushuin

University

Keiko ITO[‡]

School of Economics, Senshu University

Daisuke MIYAKAWA**

Weatherhead Center for International Affairs, Harvard University

July 2014

Abstract: This paper examines how the Japanese firms' export decision is affected by the availability of information on export markets, focusing on whether the availability of such information has a different impact on the export decision between large firms and small and medium-sized enterprises (SMEs). Unlike existing studies which solely focus on information sharing among firms, we are interested in the role of firms' lender banks as an additional source of information. Specifically, using a unique dataset containing information not only on firms' export activities but also on their lender banks' exposure to other exporting firms and lender banks' own overseas activities, we find that information spillovers through lender banks positively affect SMEs' decision to start exporting and the range of destinations to which they export. Such information spillovers also reduce the likelihood that exporter firms exit from export markets. The export-to-sales ratio of exporter firms, however, is not affected by such information spillovers. These results imply that information on foreign markets provided by lender banks substantially reduces the fixed entry costs associated with starting exporting and entering new export markets as well as firms' costs associated with continuing to export. Our results highlight that channels of information spillovers other than those examined in the literature so far may be of considerable importance, especially for SMEs.

Key words: Export Decision; Lender Bank; Information Spillover; Extensive and Intensive Margins

JEL Classification: F10, F14, G21, L25

1. Introduction

The successful globalization of Japanese firms, especially small and mediumsized enterprises (SMEs), is becoming one of the most important policy topics in Japan. Facing sluggish domestic sales against the background of an aging and shrinking population, Japanese firms have increasingly been relying on export markets for sales and profits. The share of exports in Japan's GDP has increased from 10.9% in 2000 to 14.7% in 2012. Yet, while well-established large firms have been diversifying their destinations of sales and locations abroad, it is generically difficult for SMEs to overcome the various obstacles associated with entering overseas markets. Given that SMEs account for a large share in the manufacturing sector in terms of the number of firms, the number of employees, and value added, however, it is important from a policy perspective to encourage SMEs to expand their business activities towards overseas markets. Against this background, this study examines the determinants of firms' export behavior, focusing in particular on SMEs.

The international trade literature suggests that to start exporting firms incur fixed sunk costs. These costs reflect, for example, the fact that firms initially are uncertain about their export profitability and thus have to collect a considerable amount of relevant information on export markets. Other potential costs include, for example, modifying products to suit local tastes and setting up distribution networks. Developing a theoretical model, Melitz (2003) therefore suggests that only firms which are sufficiently productive to cover such fixed costs can be exporters. Extant empirical studies (e.g., Bernard *et al.* 2003; Mayer and Ottaviano 2008; and Todo 2011) examining this hypothesis, however, indicate that there must be other important factors which affect firms' decision to export. They suggest that even when their productivity is not very high, firms can be exporters as long as other critical conditions are satisfied. This means that even SMEs, which tend to be less productive than larger firms, can potentially be exporters, and from a policy perspective, it is important to understand

what these other determinants of exporting are and what policy makers can do to help SMEs to enter export markets.

The extant literature has already focused on a number of determinants that may affect firms' export decision. One important research strand in this context concentrates on information spillovers. The underlying idea is that information exchange with other exporting firms reduces the individual fixed costs associated with exporting, and that such information exchange therefore increases the probability that a firm will export (see, e.g., Krautheim (2012) for a theoretical investigation).¹ Having access to information on foreign markets, the hypothesis goes, substantially reduces uncertainty and encourages firms to engage in export activities. Empirical work by Koenig et al. (2010) confirms this hypothesis, showing that the presence of other exporters has a positive effect on the export decision of other firms. However, although Koenig et al. (2010) find evidence of positive information spillovers, the evidence produced by other empirical studies on such information spillovers is at best weak (e.g., Aitken et al. 1997; Barrios et al. 2003; Bernard and Jensen 2004). According to a survey conducted by the Small and Medium Enterprise Agency of Japan, however, it is clear that many enterprises that would like to export face problems in terms of, e.g., "securing outstanding partner enterprises" and "ascertaining the needs of local enterprises and residents overseas." Especially compared to large enterprises, a high percentage of SMEs have not been able to undertake export operations as a result of difficulties to "secure outstanding partner enterprises." This is a serious challenge for SMEs, which have much more limited managerial resources than large enterprises (Japan Small Business Research Institute 2008). In fact, the productivity of SMEs on average is much lower than that of large firms, suggesting that many SMEs are not sufficiently profitable to afford the fixed costs of exporting. Therefore, in order for SMEs to start exporting, they have to raise their productivity or try to lower the costs of exporting. However, SMEs usually have much fewer transaction partners than large

firms. Therefore, SMEs are more likely to face serious difficulties in finding a partner enterprise abroad through information exchanges with their current transaction partners, implying that it is costly for SMEs to collect information on foreign markets and possible partner enterprises abroad. Thus, one of the most important research questions is what channels contribute to the effective information exchange between exporting firms and non-exporting firms, which is more relevant for SMEs than large firms and has not been clearly examined in extant studies. Obtaining a more detailed picture of information spillovers is important especially in the context of SMEs, since it is much less clear how such information spillovers arise for SMEs than for larger firms. For example, SMEs likely have much fewer opportunities to interact with export firms in their daily business activities than large firms.

Given these considerations on information spillovers, this study focuses on information provided by lender banks as an additional channel of information spillovers. The hypothesis we examine in this paper is that lender banks contribute to such information exchange in the form of acting as a conduit for information on export markets and potential business partners abroad. In the case of Japan, lender banks generally provide not only financial support but also business consulting services utilizing their extensive knowledge collected through lending transaction relationships and from various information sources. Since the monitoring of borrower firms is important for banks, banks in general should accumulate information on borrower firms and related parties. Thus, if we assume that a particular bank is very knowledgeable about overseas business opportunities either through its own banking activities (e.g., foreign branches) or transactions with client firms with experience in exporting, potential exporter firms would find it helpful to consult with such a bank.

The information provided by lender banks could be more important for SMEs than for large firms for the following two reasons. First, although SMEs tend to have less access to information about overseas market than larger firms (e.g., they have a smaller number of trading partners, have less exposure to overseas information through imports, and are more constrained in their internal resources for the collection of overseas market information), they usually have close ties to lender banks and therefore are in a good position to obtain feedback from banks on their business strategies. Hence, lender banks could play an important role as a conduit of export market information for SMEs. Second, lender banks themselves have a strong incentive to provide such information to client SMEs, since the expansion of client firms' business activities naturally leads to larger business opportunities for lender banks. In other words, to the extent that lender banks have accumulated overseas market information, it would be natural for them to share such information with their clients.

This study contributes to the existing literature in at least two ways. First, we examine the export decision by using a large-scale dataset that includes a large number of SMEs and makes it possible to link firm-level information with information on the lender banks of each firm. Using this dataset and focusing in particular on SMEs, we investigate whether the importance of information provided by banks is more crucial for SMEs than for large firms, given that, as mentioned above, SMEs likely find it more difficult than larger firms to collect information on export markets by themselves due to their managerial resource constraints. Specifically, to examine the role of information provided by banks and differences in this role for SMEs and large firms, we focus on whether such spillovers affect whether firms start exporting (*an extensive margin*), expand their export destinations (*another extensive margin*), stop exporting (*another extensive margin*). Second, we also consider the impact of main banks' financial health and the agglomeration of nearby exporters on firms' export decision.²

The findings suggest that information spillovers through banks positively affect SMEs' decision to start exporting and expand their export destinations. The information spillovers also reduce the likelihood of firms stopping to export. The export-to-sales ratio of exporters, however, is not affected by such information spillovers. These results imply that information on foreign markets provided by lender banks substantially reduces the fixed entry costs of exporting as well as the costs associated with maintaining firms' export status.

The remainder of this study is organized as follows. Section 2 describes the dataset used in the analysis and provides some descriptive statistics for our sample firms. Next, Section 3 briefly explains the role that main banks play in Japan and presents the empirical strategy we employ. Section 4 then presents our estimation results, while Section 5 concludes.

2. Differences in Export Behavior between Large Firms and SMEs

2.1. Data Description

Let us start by looking at differences between large firms and SMEs in terms of their export behavior and various firm characteristics. To examine firms' export behavior and various other characteristics, we use firm-level panel data for the period 1997–2008 from the *Basic Survey on Business Structure and Activities (BSBSA)* conducted annually by the Ministry of Economy, Trade and Industry (METI). The survey is compulsory and covers all firms with at least 50 employees and 30 million yen of paid-in capital in the Japanese manufacturing, mining, and wholesale and retail sectors as well as several other service sectors. The survey contains detailed information on firm-level business activities such as the 3-digit industry in which the firm operates, its number of employees, sales, purchases, exports, and imports (including a breakdown of the destination of sales and exports and the origin of purchases and imports).³ It also contains the number of domestic and overseas subsidiaries, and various kinds of financial data such as costs, profits, investment, debt,

and assets. Although the survey covers firms in the non-manufacturing sector, this study focuses on firms in the manufacturing sector only because the survey does not cover international transactions in services and only asks firms about the amount of trade in goods.⁴

The key aim of our analysis, as mentioned above, is to investigate the importance of information on destination markets and advice provided by lender banks to their client firms. To do so, we combine the firm-level data with information on firms' lender banks and examine the relationships between firm characteristics, lender banks' ability to provide advice, and firms' export status. We merge the dataset above with information on the lender banks for each firm using the information on loan relationships in Teikoku Databank Ltd.'s corporate information database. The database, called *COSMOS2*, contains the names of each firm's lender banks listed in the order of importance to the firm. (The maximum number of lender banks listed for each firm is ten.) We assume that the bank at the top of the list for a particular firm in a particular year is the main bank of that firm in that year. Information on the characteristics of lender banks' total assets, their equity ratio, and their loan-deposit ratio. We also calculate the number of client firms for each bank using our firm–bank-linked dataset. Our unbalanced panel data contain approximately 7,000 manufacturing firms each year.

2.2. Overview of the Firm-Bank-Linked Database

Using the firm–bank-linked database, we examine how large firms' and SMEs' export behavior and various other characteristics differ. We start with Table 1, which provides an overview of the share of exporters among firms of various sizes in our dataset. SMEs are defined as firms with paid-in capital of up to 300 million yen or up to 300 employees, following the Japanese legal definition of SMEs. We define all other firms in our database as large firms. We further divide SMEs into small firms and medium-sized firms in order to more closely examine differences among SMEs. Small firms are defined as firms with paid-in capital of up to 150 million yen and no more than 150 employees. All other SMEs are defined as medium-sized firms.

As shown in Table 1, the share of exporters differs considerably between large firms and SMEs. While approximately 60 percent of large firms are exporters, only 25 percent of SMEs are. The share of exporters among small firms is even smaller: 20 percent. In other words, more than 80 percent of small firms do not export. Given that nearly 90 percent of the firms in our dataset are SMEs, this means that a substantial number of manufacturing firms in our database – about 70 percent – are non-exporters.

	No. of firms	Share in all firms (%)	Share in the size category (%)
All firms	77,305	100.0	
Exporters	22,526	29.1	
Non-exporters	54,779	70.9	
Large firms	9,778	12.6	100.0
Exporters	5,876		60.1
Non-exporters	3,902		39.9
SMEs	67,527	87.4	100.0
Exporters	16,650		24.7
Non-exporters	50,877		75.3
Medium-sized firms	45,298	58.6	100.0
Exporters	12,959		28.6
Non-exporters	32,339		71.4
Small firms	22,229	28.8	100.0
Exporters	3,691		16.6
Non-exporters	18,538		83.4

Table 1: Number of firms in the dataset by size and export status

Given that far fewer SMEs are exporting than large firms, this suggests that SMEs are less likely to start exporting. In order to test this statistically, we define an export starter as a firm which did not export from year t-3 to year t-1 but exported in year t. We construct various dummy variables representing a firm's export status and examine differences in export behavior across firms of different sizes. The first dummy variable we use, NEW_EXP , takes one for firms which did not export from year t-3 to year t-1 but exported in year t. This variable takes zero for firms which did not export from years t-3 to t. Therefore, the variable NEW_EXP is not defined for firms which did export between years t-3 and t (i.e., that were always exporters). A second dummy variable we use is NEW_REGION , which takes one for firms which exported in year t-1 and increased the number of export destinations in year t. The dummy takes zero for firms which exported in year t-1 and increase the number of export destinations in year t. A third dummy variable we use is for firms that stopped

exporting, *STOP_EXP*, which takes one for firms which exported in year *t*-1 but did not export in year *t*, and zero otherwise. Finally, we construct a variable, *EXP_SALES*, which represents the ratio of exports to sales for firms which export in year *t*.

Table 2 shows the mean values for these variables. For all firms, the mean of *NEW_EXP* is 0.034, indicating that 3.4 percent of non-exporting firms in years *t*-3 to *t*-1 started exporting in year *t*. Looking at the difference between large firms and SMEs, 6.4 percent of non-exporting large firms started exporting in year *t*, while 3.2 percent of non-exporting SMEs started exporting in year *t*, and the difference in this propensity to start exporting is statistically significant at the 1 percent level. Similarly, the propensity to start exporting differs significantly between medium-sized and small firms (3.7 percent versus 2.3 percent). As for the expansion of export destinations (*NEW_REGION*), larger firms are more likely to increase their number of export destinations and the differences are statistically significant across different sizes of firms. On the other hand, smaller exporting firms are more likely to stop exporting than larger firms (*STOP_EXP*). Finally, while large firms have a significantly higher export intensity (*EXP_SALES*) than SMEs, no statistically significant difference between medium-sized and small firms (*EXP_SALES*) than SMEs, no statistically significant difference between medium-sized and small firms can be observed.

Table 2. Differences in export behavior by firm size

	All fi	rme		Large	firms vs. S	MEs		Medium-sized firms vs. Small firms						
	All I	11115	Large	firms	SM	lEs		Medium-s	ized firms	Small	firms			
Variable	Obs.	Mean	Obs.	Mean	Obs.	Mean	t-test	Obs.	Mean	Obs.	Mean	t-test		
NEW_EXP	50,385	0.034	3,711	0.062	46,674	0.032	***	29,844	0.037	16,830	0.023	***		
NEW_REGION	20,884	0.156	5,606	0.181	15,278	0.147	***	11,944	0.152	3,334	0.126	***		
STOP_EXP	20,884	0.067	5,606	0.051	15,278	0.072	***	11,944	0.066	3,334	0.093	***		
EXP_SALES	20,143	0.135	5,138	0.160	15,005	0.127	***	11,704	0.126	3,301	0.130			

Note: *** indicates that the mean values of two groups of firms are significantly different at the 1% level.

Overall, the results indicate that smaller firms are less likely to start exporting and

continue exporting. In terms of Melitz's (2003) theoretical model, these results suggest that small firms are not sufficiently productive to cover the fixed costs involved in starting and continuing to export. Thus, for small firms to start exporting, information on export markets from a variety of sources, such as nearby exporting firms, foreign investors, transaction partners, and lender banks, may play a particularly important role.

Against this background, Table 3 examines differences in firm characteristics and the availability of information on foreign markets among firms of different sizes and between exporters and non-exporters. Specifically, firm characteristics examined include firms' level of total factor productivity TFP⁵ as well as their ratio of liquid assets to total assets (F_CASH). As proxies for the availability of information on export markets, we calculate the number of nearby firms (F NEARBYFIRM and $F_NEARBYINDEXP$),⁶ the foreign ownership ratio (FOREIGN), the import ratio (IMPORTRATIO), and the share of overseas assets in total assets (FFORINV) for each firm. We also calculate several variables which proxy the amount of information on export markets provided by each firm's lender banks: the share of exporting client firms in the total number of client firms for the top-lender bank of a firm (BANKINFO), the average share of exporting client firms in the total number of client firms for all the lender banks of a firm (BANKINFO_AVR), the number of overseas branches of the top-lender bank of a firm (BANKBR), the average number of overseas branches of all the lender banks of a firm (BANKBR_AVR), and the size (total assets in logarithm) of the top-lender bank (B_SIZE). Finally, we employ a dummy variable, JBIC, which takes one for firms who borrow from the Japan Bank for International Cooperation (JBIC), formerly called the Japan Export-Import Bank. This is a government-run financial institution specializing in international financial operations such as trade finance.

Table 3 shows that, regardless of firm size, exporters tend to have higher TFP and

larger cash flow than non-exporters, and the difference in the means is significantly different in all cases. These figures indicate that exporters are more productive and less financially constrained than non-exporters, suggesting that productivity and financial resources play an important role in allowing firms to cover the fixed costs of exporting. Moreover, regardless of firm size, exporters have a larger value than non-exporters for all the variables representing availability of information on export markets or information sources. The only exception is the mean value of *JBIC* for small firms, which is larger for non-exporters, although the difference in the means is not statistically significant. All these figures indicate that exporters tend to have more information available than non-exporters, which can be interpreted as implying that utilizing such available information lowers the fixed costs of exporting.

	All f	irms	Large	firms	SN	IEs	Medium-s	ized firms	Small	firms
Variable	Exporters	Non- exporters								
TFP	0.050	-0.008	0.101	0.056	0.032	-0.013	0.043	0.000	-0.008	-0.036
F_CASH	0.580	0.542	0.537	0.507	0.595	0.544	0.593	0.541	0.602	0.550
F_NEARBYFIRM	428.008	262.111	600.068	484.389	367.286	245.064	377.732	256.626	330.610	224.894
F_NEARBYINDEXP	4.667	1.592	5.519	2.656	4.366	1.510	4.434	1.559	4.127	1.424
FOREIGN	14.565	2.160	22.328	8.533	11.825	1.671	13.465	2.192	6.068	0.763
IMPORTRATIO	0.046	0.008	0.049	0.014	0.044	0.008	0.045	0.008	0.043	0.007
FFORINV	0.033	0.004	0.051	0.009	0.026	0.003	0.028	0.004	0.020	0.002
BANKINFO	0.240	0.207	0.261	0.237	0.233	0.204	0.234	0.208	0.228	0.198
BANKINFO_AVR	0.237	0.204	0.255	0.232	0.231	0.202	0.232	0.206	0.225	0.195
BANKBR	17.932	13.863	20.354	17.580	17.077	13.578	17.451	14.219	15.764	12.461
BANKBR_AVR	15.116	12.018	16.890	14.938	14.490	11.794	14.779	12.351	13.473	10.821
B_SIZE	16.718	16.308	17.011	16.733	16.614	16.276	16.651	16.352	16.485	16.142
JBIC	0.005	0.001	0.010	0.004	0.003	0.001	0.003	0.001	0.000	0.001

Table 3. Comparison of mean values for exporters and non-exporters

Note: The difference between exporters and non-exporters is statistically significant at the 1% level in all cases except for *JBIC* for small firms.

An interesting finding is that the average TFP for exporting SMEs is much lower than the average TFP for non-exporting large firms (0.032 vs. 0.056). On the other hand, regarding the mean values for the information-related variables, the difference between exporting SMEs and non-exporting large firms looks small. Exporting SMEs tend to have larger mean values than non-exporting large firms for proxies representing information directly obtained by firms themselves (*FOREIGN*, *IMPORTRATIO*, *FFORINV*) rather than via intermediaries such as nearby firms or banks. As for proxies of information provided by a firm's lender banks, although the mean values for exporting SMEs are smaller than those for non-exporting large firms, the difference is quite small.

Thus, it appears that SMEs are inferior to large firms in terms of both firms' own performance and the availability of various information sources. Even exporting SMEs are less productive than non-exporting large firms. The findings thus suggest that, on their own, many SMEs are unlikely to be sufficiently productive to cover the costs of exporting and therefore need to rely on a variety of information sources to lower such costs and overcome the barriers to exporting. Thus, the results thus far allow us to conjecture that the availability of a variety of information sources is more critical for SMEs to start exporting than for large firms.

3. Empirical Strategy

3.1. The Main Bank System in Japan

Before turning to our statistical analysis, let us consider the various roles banks play in Japan, given that we focus on one of these roles here, namely their role as information provider. Traditionally, lender banks in Japan, particularly main banks, provide not only financial services but also various consulting services to their client firms. In this study, we focus on the importance of the latter role. The so-called "main bank system" has been a key feature of Japan's economic system that can be traced back as far as the early post-war period.⁷ In this system, a firm's "main bank" usually is the bank from which it has borrowed the most and with which it typically has a longterm relationship. In addition, it is widely argued that main banks not only provide loans to client firms but also play a consulting role by providing relevant business information. In addition, main banks may get involved in the management of a firm in times of distress. Although the extent and form of main banks' involvement in firms' management in times of financial difficulties have been changing over time, main banks are still perceived to play an important role as providers of both funds and information to their client firms.

Trying to provide a theoretical underpinning for such long-term relationships between main banks and borrower firms, Patrick (1994) argues that such relationships enable banks to gain access to "soft information" on borrower firms, which helps to raise the efficiency of loan screening and borrower monitoring. The argument that repeated bank loan transactions lead to the accumulation of soft information on client firms has also been voiced in more recent studies such as Degryse *et al.* (2009).

Such soft information on borrower firms and banks' own ability to collect information on industry-, region-, and nation-wide businesses has helped Japanese main banks to provide effective and useful financial and consulting services to their client firms, and thereby has contributed both to main banks' profits and the growth of their client firms' business. Particularly in recent years, aware of the fact that the growth prospects for Japan's domestic market are not necessarily promising and domestic manufacturing production has in fact been shrinking, banks have been promoting various services to support client firms' international activities. With more and more large Japanese firms relocating production overseas, smaller domestic firms have been forced to reduce output, resulting in sluggish demand for bank lending, which in turn has reduced business opportunities for banks in Japan. Moreover, if banks' existing client firms went out of business, banks would not only lose their current business but also future business in which to utilize the firm-specific soft information they have accumulated. Thus, faced with a potentially shrinking market at home, many banks in recent years have put greater emphasis on providing support services for client firms seeking to exploit growth opportunities overseas.

Concrete examples of such kind of support services that banks provide to their borrowers to help them with regard to international activities are provided by a Japanese Bankers Association (JBA) report (Japanese Bankers Association 2011). According to the report, other than traditional banking services such as the usual loan business, deposit services, payment services, lease and leaseback deals, or the issue of stand-by letters of credit, main banks often provide client firms with information on potential business partners overseas as well as advice on employee recruitment, advertising, tax systems, and administrative issues such as accounting systems, laws, and regulations. These examples indicate that banks provide not only financial transactions but also information services. In the report, the JBA cites a survey it conducted according to which 38 out of 43 Japanese banks with activities in Asia say they provide services other than loan, deposit, and payment services. Specifically, 32 out of the 38 banks with activities in Asia say they provide information related to investment (i.e., tax and accounting systems, etc.), while 31 banks provide opportunities for business matching (e.g., organizing business matching events for Japanese firms and potential local partners). In addition, many banks provide information on firms located in destination regions (14 banks), loan guarantees (12 banks), and support with export and import procedures (8 banks).

3.2 Empirical Model

This section explains the empirical strategy we employ to investigate the determinants of export dynamics. We are particularly interested in the impact of information provided by main banks on firms' export dynamics as represented by (i) the initiation of exports, (ii) the expansion of export destinations, (iii) the termination of exports, and (iv) the intensity of exports. To examine the first three, i.e., (i) to (iii), concerning the extensive margin, we focus on the probabilities that a firm starts

exporting, expands its export destinations, or stops exporting, while we use the exportto-sales ratio to represent the intensive margin of exports (iv).

Following previous empirical studies on the determinants of the extensive margin (e.g., Koenig *et al.* 2010, Minetti and Zhu 2011), we assume that firm *i* starts exporting, expands its export destinations, or stops exporting if its profits are larger when doing so than when not doing so. Let π_{itl}^*, π_{it2}^* , and π_{it3}^* respectively represent the difference between the profits of firm *i* when it starts exporting, expands its export destinations, or stops exporting at time *t* and its profits when not doing so. The differences are determined by the firm's characteristics, its financial condition, the characteristics of its main bank, and the availability of information on export markets to the firm. The availability of information on export, expanding export destinations, or continuing to export, and hence to lower either the variable or the fixed costs associated with exporting. While information spillovers from nearby exporter firms are also taken into account, we are particularly interested in information provided through the main bank and the non-main banks of the firm. Therefore, we parameterize π_{itk}^* as follows:

$$\pi_{itk}^* = \alpha_{1k} + Z_{it-1}\beta_{1k} + I_{it-1}\gamma_{1k} + \varepsilon_{it} \Box \hbar \Box \Box \Box \Box = 1, 2, 3$$

$$(1)$$

 Z_{it-1} is a vector of controls for firm characteristics and firms' financial condition which may affect their differential profits π_{itk}^* ; I_{it-1} is a vector of variables representing information available to firms; and ε_{it} captures unobserved firm characteristics and other unknown factors that may also affect differential profits.

We assume that firm *i* starts exporting, expands its export destinations, or stops exporting if differential profits $\pi_{itk}^*>0$. Under the assumption that ε_{it} is a normally distributed random error with zero mean and unit variance, the probabilities that firm *i* starts exporting, expands its export destinations, or continues exporting can be written

as follows:

$$\operatorname{Prob}_{itk} = \operatorname{Prob}(\alpha_{1k} + Z_{it-1}\beta_{1k} + I_{it-1}\gamma_{1k} + \varepsilon_{it} > 0) \Box h \Box \Box \Box = 1, 2, 3$$
(2)

In the first instance, we estimate equation (2) with a random effect panel probit approach. In order to take any potential endogeneity into account, we lag all right-hand side variables by one year. The dependent variable $Prob_{itk}$ denotes the change in export status at the firm-level and takes a value of 1 if a firm starts exporting at time t (k=1), increases the number of export destinations at time t (k=2), or stops exporting at time t (k=3). We define a firm as an export starter ($Prob_{itl}$ takes a value of one) if it did not export during the preceding three years (i.e., from t-3 to t-1) but exports at time t. On the other hand, $Prob_{itl}$ takes a value of zero if a firm did not export during the three years prior to year t and does not export in year t. Firms which always export are not included in the estimation of $Prob_{itl}$. For the estimation of $Prob_{it2}$ and $Prob_{it3}$, we only use firms which exported at t-1.

For the intensive margin, we estimate the following model using panel estimation assuming firm-level fixed effects:

$$EXP_SALES_{it} = \alpha_{14} + Z_{it-1}\beta_{14} + I_{it-1}\gamma_{15} + \eta_i + \varepsilon_{it}$$
(3)

where the dependent variable EXP_SALES_{it} denotes the ratio of exports to total sales measured at the firm level. For this estimation, we only use firms which exported at time *t*. Regarding control variables for firm characteristics and firms' financial condition (Z_{it}), we include the TFP level of the firm, which is measured using the multilateral TFP index method developed by Good *et al.* (1997). Based on the results of both theoretical and empirical studies, we expect firms' decision to start exporting to be positively correlated with their TFP level. Further, to take the impact of liquidity constraints on firms' export behavior into account, we include a variable representing firms' financial characteristics, namely, the ratio of liquidity assets to total assets (F_CASH) . The reason for including this variable is that, as highlighted by, e.g., Manova *et al.* (2011), Feenstra *et al.* (2013), and Minetti and Zhu (2011), financial constraints are likely to prevent firms from exporting because firms need sufficient liquidity in order to meet the entry costs associated with starting exporting. Therefore, we expect that firms with more liquidity are more likely to start exporting.

We also control for the financial health of main banks. Feenstra *et al.* (2013), for example, find that the health of banks providing trade finance is an important determinant of firm level exports. As proxy variables for main banks' financial health, we employ bank size (the log of the total assets of the bank, B_SIZE), the equity ratio (B_CAP), and the loan-deposit ratio (B_LTD).

Regarding information available to a firm (I_{ii}), we include variables representing the availability of information on export markets accumulated by lender banks (i.e., both main and non-main banks) as well as information spillovers from nearby firms. The explanatory variables of main interest are those representing information on export markets potentially available to a firm through its main bank and other lender banks, which are proxies for the amount of information firm *i*'s main bank and other lender banks have accumulated. Specifically, to gauge the information potentially available to a firm, we use (i) the ratio of the number of exporting client firms to the total number of the main bank's client firms, i.e., the intensity of each main bank's dealings with exporting firms, *BANKINFO*; (ii) the average of the same variable for all the lender banks, *BANKINFO_AVR*; (iii) the number of foreign branches of the main bank, *BANKBR*; and (iv) the average of the same variable for all the lender banks, *BANKBR_AVR*.

We conjecture that banks which have a higher share of exporters in their total number of client firms (*BANKINFO* or *BANKINFOR_AVR*) and/or have a larger number of overseas branches (*BANKBR*) likely accumulate more information on overseas markets. For instance, if banks' allocation of internal managerial resources is related to the importance of a particular type of lending activity in the banks' overall business, we would expect banks that have extensive dealings with exporting firms to devote more management resources to such lending activity, including the gathering of overseas market information, etc.

Note that using the averages of the variables representing lender banks' information is likely to mitigate potential bias from any systematic matching between firms and main banks. Suppose, for example, that firms that are about to start exporting tend to choose a bank with more information on export markets. If this is the case, causality would run from firms' export decision to the main bank information variables, which would give rise to bias in our estimation of the coefficients associated with the main bank information variables. Given that it is very unlikely that all the major lender banks listed in the *COSMOS2* database for a particular firm will change completely just because the firm is planning to start exporting, using the averages of the lender bank information variables should mitigate the potential endogeneity from this reverse causality. Thus, the purpose of estimating the empirical model using the averages of the information variables across all banks that a firm borrows from instead of just its main bank is to check the robustness of our results.

Given that our main interest is the role of information spillovers from lender banks to SMEs, we run the regressions based on equations (2) and (3) for the sample as a whole and separately for subsamples consisting of large firms and SMEs. In addition, to examine the importance of information spillovers for SMEs in more detail, we further divide SMEs into small and medium-sized firms and again run regressions based on equations (2) and (3).

In addition to information provided by banks, information spillovers from nearby exporters may also play a role. In order to examine whether this is the case, we include the two variables for nearby firms, $F_NEARBYFIRM$ and $F_NEARBYINDEXP$,

defined in Section 2.2. Industry dummies (for fifteen manufacturing industries) and year dummies are also included in order to control for industry- and time-specific fixed effects. Summary statistics for all the variables used in our empirical analysis and the distribution of *BANKINFO* over banks in our dataset in the year 2000 are provided in Appendixes 1 and 2, respectively.

4. Estimation Results

Tables 4 to 7 summarize the results of our estimation of the extensive and intensive margins of exports based on equations (2) and (3). In each table, the columns labeled (a), (b), and (c) show the estimation results for the sample of all firms, large firms, and SMEs, respectively. Further, columns (c1) and (c2) show the results for medium-sized and small firms, respectively. For each size category, columns (i) and (ii) respectively show the results using the main bank information variables and those using the average of the information variables across all of a firm's lender banks.

Starting with the results regarding whether firms start exporting (*NEW_EXP*), Table 4 shows that the extent to which lender banks transact with other exporting firms (*BANKINFO* or *BANKINFO_AVR*) plays an important role in whether firms start exporting. However, examining the role of such information spillovers from lender banks for SMEs and large firms separately shows that they have a strong positive impact in the case of the former but not in the case of the latter. This implies that the information provided by banks is an important determinant of whether SMEs start exporting. Moreover, this result holds regardless of whether we focus on information spillovers from the main bank only or for all lender banks of a firm (columns (i) and (ii)), confirming the robustness of the estimation results. In line with the discussion above, the results obtained can be interpreted as indicating that, in contrast with larger firms, SMEs lack the internal resources and access to other external information to

gather sufficient information on overseas markets on their own, so that lender banks appear to be important sources of such information, which enables firms to become exporters.

Next, the number of lender banks' overseas branches (BANKBR) or BANKBR_AVR) also has a positive impact on whether firms start exporting. One interesting difference between large firms and SMEs is that for the former the average number of lender banks' overseas branches (BANKBR_AVR) matters, while for the latter only the main bank's number of overseas branches (BANKBR) matters. Given that banks' overseas branches play an important role in processing client firms' overseas payments, a possible interpretation of this difference is that SMEs solely rely on their main bank to process overseas transactions, while large firms tend to use the overseas branches of several lender banks rather than using their main bank only. When SMEs start exporting, the transaction volume involved is likely to be relatively small and their overseas financial transactions can be handled by their main bank. On the other hand, when large firms start exporting, they are likely to have a larger number of transaction partners overseas, so that they may need to utilize a greater number of overseas branches in numerous countries. This may be the reason for the significant positive coefficient on BANKBR_AVR for large firms.

Turning to the results for TFP, this, as predicted by theory, has a positive impact on whether firms start exporting (see column (a)). However, in the various subsamples, the coefficient on TFP is not significant, suggesting that the impact of TFP to a great extent overlaps with the impact of firms' size. That is, the significant positive coefficient in column (a) is explained by the fact that larger firms tend to have higher TFP than smaller firms, but TFP differences within each subsample do not play a significant role in explaining whether a firm starts exporting or not. Further, also as predicted by theory, firms' liquidity (F_CASH) has a positive impact on whether they start exporting. Interestingly, this matters only for SMEs but not for large firms. As exporting is riskier than selling products domestically, firms need to have sufficient cash flow in order to take the risk of starting exporting, and the result suggests that particularly for SMEs, for which fund-raising or borrowing is usually more difficult than for large firms, having sufficient liquidity is an important determinant of whether they can deal with the uncertainties of entering foreign markets.

Looking at the results for other variables, we find that firms with a higher overseas investment ratio (*FFORINV*) or import ratio (*IMPORTRATIO*), which we use as proxies for the degree of firms' exposure to overseas markets, are more likely to start exporting. On the other hand, information spillovers from nearby firms or nearby exporters (*F_NEARBYFIRM* or *F_NEARBYINDEXP*) do not seem to play a significant role. This finding conflicts with the result obtained by Koenig *et al.* (2010), but is consistent with the results found in several other studies, including Aitken *et al.* (1997), Barrios *et al.* (2003), and Bernard and Jensen (2004). Finally, banks' balance sheet variables (i.e., *B_SIZE*, *B_CAP*, and *B_LTD*) also do not have any significant impact on whether firms start exporting.

Random-effect panel logit		(a) All	firms																	
						(b) Larg	ge firms			(c) S	MEs									
Dependent variable:														(c1) Mediun	n-sized firms	8		(.)	all firms	
NEW_EXP(t)		in bank	()	verage		in bank	()	verage		in bank	()	verage		in bank	()	verage		ain bank	()	verage
	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.
BANKINFO (t-1)	2.070	0.614 ***			-1.761	1.673			2.624	0.671 ***			2.384	0.779 ***			2.971	1.404 **		
BANKINFO_AVR (t-1)			3.480	0.765 ***			-2.664	2.189			4.419	0.838 ***			3.946	0.982 ***			5.252	1.720 ***
BANKBR (t-1)	0.006	0.003 **			0.005	0.007			0.007	0.003 **			0.007	0.003 **			0.006	0.006		
BANKBR_AVR (t-1)			0.002	0.004			0.022	0.010 **			-0.002	0.004			-0.001	0.005			-0.007	0.009
FFORINV (t-1)	6.213	1.060 ***	6.136	1.057 ***	7.651	2.767 ***	7.729	2.804 ***	5.684	1.178 ***	5.609	1.177 ***	2.919	1.427 **	2.897	1.429 **	10.731	2.087 ***	10.545	2.079 ***
JBIC (t-1)	0.189	0.802	0.106	0.797	0.263	1.326	0.219	1.350	-0.098	1.085	-0.247	1.085	-0.074	1.099	-0.182	1.099	-17.566	1.8E+04	-18.947	2.9E+04
B SIZE (t-1)	-0.052	0.038	0.005	0.031	0.012	0.108	-0.061	0.096	-0.082	0.041 **	0.008	0.034	-0.079	0.048 *	0.012	0.040	-0.115	0.083	-0.017	0.068
B CAP (t-1)	-4.072	3.090	-2.792	3.021	-12.629	9.742	-13.638	9.530	-3.118	3.294	-1.496	3.223	-2.655	3.879	-0.993	3.792	-3.335	6.515	-1.599	6.371
B_LTD (t-1)	-0.172	0.217	-0.151	0.210	-0.058	0.409	-0.170	0.400	-0.233	0.265	-0.199	0.256	-0.322	0.290	-0.290	0.277	-0.137	0.741	-0.073	0.733
F_NEARBYFIRM (t-1)	1.0E-04	9.2E-05	7.3E-05	9.2E-05	2.6E-04	2.1E-04	2.8E-04	2.1E-04	-4.5E-05	1.1E-04	-7.7E-05	1.1E-04	-3.2E-05	1.3E-04	-5.5E-05	1.3E-04	-1.9E-04	2.6E-04	-2.4E-04	2.6E-04
F_NEARBYINDEXP (t-1)	0.009	0.007	0.010	0.007	0.009	0.011	0.008	0.011	0.010	0.009	0.010	0.009	0.008	0.010	0.008	0.010	0.028	0.024	0.029	0.024
IMPORTRATIO (t-1)	3.628	0.583 ***	3.604	0.583 ***	3.256	1.501 **	3.366	1.513 **	3.873	0.643 ***	3.875	0.644 ***	3.742	0.765 ***	3.745	0.764 ***	5.502	1.339 ***	5.455	1.331 ***
FOREIGN (t-1)	0.001	0.001 *	0.001	0.001 *	0.000	0.002	0.000	0.002	0.001	0.001	0.001	0.001 *	0.001	0.001	0.001	0.001	0.001	0.003	0.001	0.002
TFP (t-1)	0.584	0.280 **	0.543	0.280 *	0.154	0.909	0.138	0.923	0.358	0.302	0.316	0.303	-0.068	0.351	-0.103	0.351	0.562	0.657	0.504	0.657
F_CASH (t-1)	0.504	0.247 **	0.512	0.247 **	1.175	0.773	1.166	0.783	0.637	0.267 **	0.647	0.268 **	0.540	0.308 *	0.544	0.308 *	1.385	0.577 **	1.407	0.577 **
Firm random effect	Y	'es	Y	es	Y	es	Y	es	Y	es	Y	es	Y	es	Y	'es	Y	l'es	Y	l es
Industry effect	Y	'es	Y	'es	Y	es	Y	'es	Y	'es	Y	'es	Y	'es	Y	'es	N N	<i>l</i> es	Y	<i>l</i> es
Year effect	Y	'es	Y	'es	Y	es	Y	'es	Y	'es	Y	es	Y	'es	Y	es	Y Y	<i>l</i> es	Y	<i>l</i> es
Number of obs.		37,7	798			2,7	70			35,	028			22,	507			12,	521	
Number of groups		9,3	70			81	5			8,8	303			6,0	013			3,7	762	
Obs. per group: Min		1				1	l			1	l			1	1				1	
Average		4				3.	4			4	1			3.	.7			3	.3	
Max		1	D				0			1	0			1	0			1	0	
Wald chi2		1.34		4.7		.11	-	8.3		7.44		0.82	-	4.58		5.19		ə.78		2.66
Prob > chi2		0000		0000		018		826	0.0			0000		0000		0000		0000		0000
Log likelihood	-41	96.0	-41	94.0	-49	91.4	-48	39.2	-36	62.3	-36	60.2	-26	80.8	-26	80.8	-9	33.9	-9	32.2
Likelihood-ratio test of rho=0																				
Chibar2		.41		.63		18		.63		3.76		.44		.66		.90		.70		.48
Prob >= chibar2	0.0	0000	0.0	0000	0.0	110	0.0	090	0.0	0000	0.0	0000	0.0	0000	0.0	0000	0.0	0030	0.0	0000

Table 4: Estimation results for NEW_EXP

Note: ***, **, and * indicate significance at the 1, 5, and 10% level, respectively.

Next, let us consider the quantitative implications of our estimation results. Specifically, let us focus on the results for the SME subsample (i.e., column (c)). The estimated marginal effects of *BANKINFO* and *BANKINFO_AVR* are 2.624 and 4.419, respectively. Suppose that for the subsample of non-exporter SMEs these variables representing information spillovers through lender banks increase by one standard deviation (i.e., 0.073 and 0.056 in panel (c) in Appendix 1) in year *t-1*. Given the estimated marginal effects, the model predicts that the probabilities that a non-exporter SME will start exporting are $2.642 \times 0.073 = 19.3$ percentage points and $4.419 \times 0.056 = 24.7$ percentage points higher than in the case that there is no change in the banks information variables. Considering that the sample mean and the standard deviation of the probability that an SME will exporting are 3.2% and 17.6% respectively, this implies that information spillovers through lender banks have an economically sizable impact on whether firms will start exporting.

Let us now examine the estimation results when focusing on the expansion of export destinations (*NEW_REGION*), which are shown in Table 5. Information spillovers from lender banks have a positive and significant impact, particularly for SMEs, which is consistent with the results for *NEW_EXP* shown in Table 4. This means that the information provided by lender banks plays an important role not only for the initiation of exports but also for the expansion of export destinations. We further find that firms' liquidity (*F_CASH*) has a positive impact on the expansion of export destinations, which is also consistent with the results for *NEW_EXP*. However, while *F_CASH* does not have a statistically significant impact for large firms when focusing on *NEW_EXP*, it does have a positive and significant impact for large firms in the case of *NEW_REGION*, suggesting that even for large firms expanding export destinations requires a certain level of liquidity. Finally, information spillovers from nearby firms have a positive, although only weakly significant, impact on the expansion of destinations in the full sample consisting of firms of all sizes (see column (a)).

Random-effect panel logit		(a) All	firms																	
						(b) La	rge firms			(c) S	MEs									
Dependent variable:														(c1) Mediu	m-sized firms	8		()	nall firms	
NEW_REGION(t)	(i) Main b		(ii) A	verage	(i) Ma	in bank	(ii) A	verage	() ···	in bank	()	verage	(i) Ma	in bank	()	verage		in bank	(ii) A	verage
	dy/dx Std	. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.
BANKINFO (t-1)	1.393	0.390 ***			0.733	0.633			1.536	0.502 ***			1.494	0.551 ***			2.074	1.226 *		
BANKINFO_AVR (t-1)			1.580	0.469 ***			0.489	0.801			1.820	0.589 ***			1.858	0.652 ***			1.747	1.419
BANKBR (t-1)	0.002	0.001			-0.001	0.002			0.004	0.002 **			0.005	0.002 **			0.003	0.005		
BANKBR_AVR (t-1)			0.003	0.002			0.007	0.004 *			0.001	0.003			0.001	0.003			0.001	0.006
FFORINV (t-1)	-0.646	0.436	-0.664	0.436	-0.518	0.704	-0.554	0.705	-0.850	0.568	-0.859	0.568	-0.572	0.601	-0.567	0.602	-2.769	1.583 *	-2.815	1.582 *
JBIC (t-1)	0.349	0.305	0.272	0.306	0.473	0.355	0.460	0.358	-0.270	0.599	-0.375	0.599	-0.340	0.596	-0.450	0.597	N.A.	N.A.	N.A.	N.A.
B SIZE (t-1)	0.008	0.024	0.024	0.020	0.055	0.045	0.000	0.039	-0.032	0.028	0.020	0.024	-0.033	0.032	0.026	0.027	-0.035	0.066	0.006	0.055
B CAP (t-1)	-1.415	1.991	-1.095	1.944	1.422	3.878	-0.325	3,761	-2.173	2.344	-1.187	2.296	-2.215	2.626	-1.003	2.566	-3.017	5.324	-2.658	5.267
B_LTD (t-1)	0.104	0.100	0.135	0.097	-0.120	0.142	-0.067	0.135	0.258	0.144 *	0.265	0.140 *	0.258	0.149 *	0.260	0.145 *	0.191	0.552	0.229	0.548
F NEARBYFIRM (t-1)	9.7E-05 5.5	5E-05 °	1.0E-04	5.5E-05 °	1.3E-04	8.9E-05	1.3E-04	8.9E-05	1.7E-05	7.0E-05	2.0E-05	7.0E-05	-1.0E-05	7.5E-05	-6.4E-06	7.5E-05	1.1E-04	2.0E-04	1.3E-04	2.0E-04
F_NEARBYINDEXP (t-1)	0.000	0.004	0.000	0.004	-0.003	0.009	-0.002	0.009	0.002	0.005	0.002	0.005	0.004	0.005	0.004	0.005	-0.011	0.018	-0.011	0.018
IMPORTRATIO (t-1)	0.016	0.279	-0.001	0.279	-0.112	0.526	-0.133	0.525	0.009	0.339	-0.005	0.340	-0.351	0.387	-0.368	0.388	1.485	0.763 *	1.490	0.765 *
FOREIGN (t-1)	0.000	0.000 *	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.001	0.001	0.001	0.001
TFP (t-1)	0.212	0.170	0.200	0.170	-0.430	0.318	-0.441	0.317	0.232	0.209	0.221	0.209	0.255	0.232	0.240	0.232	-0.230	0.502	-0.221	0.502
F_CASH (t-1)	0.435	0.187 **	0.423	0.187 **	1.145	0.351 ***	1.153	0.350 ***	0.425	0.229 *	0.416	0.229 *	0.356	0.257	0.350	0.257	0.825	0.517	0.800	0.518
Firm random effect	Yes		Y	'es	Y	es	Y	es	Y	es	Y	'es	Y	'es	Y	'es	Y	'es	Y	es
Industry effect	Yes		Y	'es	Y	es	Y	es	Y	es	Y	es	Y	es	Y	es	Y	es	Y	es
Year effect	Yes		Y	es	Y	es	Y	es	Y	es	Y	'es	Y	'es	Y	es	Y	es	Y	es
Number of obs.		19,9	42			5,	406			14,	536			11	,367			3,	169	
Number of groups		4,78	30			1,	245			3,8	316			3,	064			1,	077	
Obs. per group: Min		1					1				1				1				1	
Average		4.2	2			4	4.3			3	.8			3	3.7			2	2.9	
Max		10)				10			1	0				10				10	
Wald chi2	163.29	Ð	160	0.48	83	.94	86	.53	124	4.25	11	6.71	115	5.27	10	8.65	41	.89	39	.95
Prob > chi2	0.0000)	0.0	0000	0.0	000	0.0	000	0.0	000	0.0	000	0.0	0000	0.0	0000	0.1	966	0.2	596
Log likelihood	-8405.8	8	-84	07.3	-24	84.6	-24	83.4	-58	77.5	-58	81.4	-47	09.0	-47	12.4	-11	46.8	-11	47.9
Likelihood-ratio test of rho=0																				
Chibar2	271.33	3	273	3.28	29	.16	28	.69	246	5.21	24	9.96	17	1.58	17-	4.79	62	2.58	62	.54
Prob >= chibar2	0.0000)	0.0	0000	0.0	000	0.0	000	0.0	000	0.0	000	0.0	0000	0.0	0000	0.0	0000	0.0	000

Table 5: Estimation results for NEW_REGION

Note: ***, **, and * indicate significance at the 1, 5, and 10% level, respectively.

Next, Table 6 shows the results for the exit from exporting. They indicate, first, that for SMEs, *BANKINFO_AVR* lowers the probability that a firm will stop exporting. In other words, firms are more likely to continue exporting when their lender banks have more information on export markets, implying that maintaining relationships with such lender banks reduces firms' recurring fixed costs associated with exporting, such as updating information on overseas markets.⁸ Second, in contrast with the results for *NEW_EXP* and *NEW_REGION*, in the estimation here main banks' loan-to-deposit ratio (*B_LTD*) plays a significant role, lowering the likelihood that firms exit from exporting. This means that firms whose main bank provides loans more readily are more likely to remain exporters. A possible interpretation is that such firms may find it easier to borrow funds from their main bank for trade financing and other export-related expenses and therefore are more likely to continue exporting. And third, a higher *FFORINV* and a higher *IMPORTRATIO* significantly reduce the probability that firms will stop exporting, suggesting that a firm's own international transactions such as foreign investments and imports help the firm to continue exporting.

Random-effect panel logit		(a) Al	l firms																	
						(b) La	ge firms			(c) S	MEs		_							
Dependent variable:														(c1) Medium	n-sized firms	;		()	nall firms	
STOP_EXP(t)	(i) Ma	in bank	(ii) A	verage	(i) Ma	in bank	(ii) A	verage	(i) Ma	in bank	(ii) A	verage	(i) Ma	in bank	()	verage	(i) Ma	in bank	(ii) A	verage
	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.
BANKINFO (t-1)	-0.334	0.642			-1.571	1.204			0.016	0.794			0.056	0.931			-0.339	1.603		
BANKINFO_AVR (t-1)			-1.491	0.795 *			-0.182	1.519			-2.227	0.974 **			-2.588	1.163 **			-1.186	1.840
BANKBR (t-1)	-0.002	0.002			-0.002	0.005			-0.002	0.003			0.000	0.003			-0.005	0.006		
BANKBR_AVR (t-1)			-0.006	0.004			-0.008	0.008			-0.004	0.004			0.002	0.005			-0.019	0.009 **
FFORINV (t-1)	-4.426	0.863 ***	-4.357	0.862 ***	-3.042	1.570 *	-3.039	1.575 °	-4.765	1.066 ***	-4.705	1.062 ***	-4.716	1.249 ***	-4.674	1.246 ***	-4.419	1.988 **	-4.450	2.003 **
JBIC (t-1)	-0.697	0.806	-0.609	0.806	-0.785	1.154	-0.787	1.155	-0.566	1.174	-0.459	1.174	-0.422	1.190	-0.311	1.193	N.A.	N.A.	N.A.	N.A.
B SIZE (t-1)	0.009	0.039	0.011	0.033	-0.020	0.089	-0.004	0.078	0.024	0.044	0.024	0.037	-0.001	0.053	-0.004	0.044	0.061	0.086	0.088	0.070
B CAP (t-1)	5.853	3.120 *	5.559	3.052 *	12.594	7.316 *	13.155	7.183 *	4.575	3,552	4.132	3.479	1.748	4.246	1.510	4.145	10.718	6.731	10.308	6.682
B_LTD (t-1)	-0.489	0.225 **	-0.469	0.221 **	-0.136	0.327	-0.198	0.322	-0.689	0.323 **	-0.623	0.316 **	-1.230	0.442 ***	-1.154	0.434 ***	0.483	0.597	0.543	0.593
F NEARBYFIRM (t-1)	-8.8E-05	1.1E-04	-6.6E-05	1.1E-04	4.5E-05	2.1E-04	6.0E-06	2.1E-04	-3.1E-05	1.4E-04	8.4E-06	1.3E-04	6.7E-05	1.5E-04	1.1E-04	1.5E-04	-3.9E-04	3.0E-04	-3.7E-04	3.0E-04
F_NEARBYINDEXP (t-1)	-0.011	0.010	-0.010	0.010	0.015	0.020	0.015	0.020	-0.023	0.013 *	-0.022	0.013 *	-0.028	0.015 *	-0.028	0.015 *	0.002	0.026	0.002	0.026
IMPORTRATIO (t-1)	-1.054	0.515 **	-1.055	0.515 **	-1.599	1.269	-1.583	1.276	-1.056	0.583 *	-1.070	0.582 *	-0.600	0.672	-0.628	0.672	-2.576	1.211 **	-2.465	1.216 **
FOREIGN (t-1)	0.000	0.001	0.000	0.001	0.000	0.001	0.000	0.001	0.000	0.001	0.000	0.001	0.000	0.001	0.000	0.001	-0.011	0.011	-0.011	0.011
TFP (t-1)	-0.819	0.285 ***	-0.787	0.285 ***	-0.413	0.662	-0.436	0.663	-0.628	0.335 *	-0.592	0.334 *	-0.837	0.398 **	-0.804	0.398 **	0.868	0.645	0.870	0.649
F_CASH (t-1)	0.045	0.316	0.061	0.316	-0.031	0.746	-0.040	0.747	-0.124	0.367	-0.094	0.367	-0.046	0.441	-0.029	0.441	-0.660	0.671	-0.577	0.675
Firm random effect	Y	'es	Y	'es	Y	es	Y	'es	Y	'es	Y	'es	Y	'es	Y	es	Y	/es	Y	es
Industry effect	Y	es	Y	es	Y	es	Y	es	Y	es	Y	es	Y	'es	Y	es	Y	/es	У	'es
Year effect	Y	'es	Y	'es	Y	es	Y	'es	Y	'es	Y	'es	Y	'es	Y	es	Y	les	Y	'es
Number of obs.		19,	942			5,	406			14,	536			11,3	367			3,	169	
Number of groups		4,7	780			1,	245			3,8	316			3,0)64			1,	077	
Obs. per group: Min			1				1			1	l			1	l				1	
Average		4.	.2			4	1.3			3.	.8			3.	.7			2	2.9	
Max		1	0				10			1	0			1	0				10	
Wald chi2	19	1.07	19	5.62	57	.94	56	5.58	152	2.10	15	7.77	12	3.6	12	7.82	60).38	63	.25
Prob > chi2	0.0	000	0.0	000	0.0	117	0.0	158	0.0	000	0.0	000	0.0	000	0.0	000	0.0	0049	0.0	024
Log likelihood	-45	51.2	-45	48.6	-99	97.0	-99	97.7	-35	19.6	-35	16.5	-25	70.7	-25	68.1	-9	15.4	-9	12.9
Likelihood-ratio test of rho=0																				
Chibar2	275	5.18	27	5.03	66	.92	67	.76	217	7.45	215	5.00	170	0.58	169	9.54	34	4.93	35	.66
Prob >= chibar2	0.0	000	0.0	000	0.0	000	0.0	000	0.0	000	0.0	000	0.0	000	0.0	000	0.0	0000	0.0	0000

Table 6: Estimation results for STOP_EXP

Note: ***, **, and * indicate significance at the 1, 5, and 10% level, respectively.

Finally, Table 7 presents the estimation results for the intensive margin (i.e., the export-to-sales ratio, EXP_SALES). The table shows the following. First, the most important finding is that although, as seen above, information from lender banks has a positive effect on firms' extensive margin of exports (starting and stopping exporting and expanding export destinations), it does not have any significant impact on firms' intensive margin of exports. This result implies that the information provided by banks mainly reduces the fixed costs (rather than the variable costs) associated with exporting. Second, among the independent variables, firms' overseas investment (FFORINV) shows a negative impact on the intensive margin. This suggests that own exporting activities and overseas production may be substitutes. Third, the foreign ownership ratio (FOREIGN) in most cases has a positive and significant impact on the intensive margin of exports, suggesting that foreign participation tends to increase export intensity. This result indicates that cooperation with foreign parent firms or investors in export markets plays an important role. However, in the case of small firms, FOREIGN has a significant negative impact (column (c2)). A possible reason is that the aims or characteristics of foreign investors investing in small firms differ from those of investor in large firms. This is an interesting result that warrants closer investigation, which, unfortunately, is beyond the scope of the present study. Fourth, while transacting with JBIC did not have any significant effect on the extensive margins (Tables 4 to 6), it does have a weakly significant positive effect on the intensive margin in the sample consisting of all firms. This result suggests that JBIC helps to increase the exports of Japanese firms by financing their export activities, and that this provision of financing from JBIC lowers the variable costs incurred by exporting firms. This is consistent with the fact that firms usually consult with JBIC with regard to trade financing once they have decided to start exporting rather than before deciding to do so. However, the variable JBIC is insignificant in all the estimation results using subsamples (columns (b) and (c)). This suggests that the effect

of transacting with JBIC more or less overlaps with the effect of firm size, but another reason may be that only a small number of exporters (especially SMEs) borrow from JBIC. As shown in Table 3 above, only 0.5 percent of exporters report that JBIC is one of their top-ten lender banks. Nevertheless, the results in Table 7 indicate that JBIC plays some part in promoting and increasing exports by Japanese firms.

Fixed-effect panel estimation		(a) All fi	ïrms																	
						(b) Larg	ge firms			(c) S	SMEs									
Dependent variable:														(c1) Medium				()	nall firms	
EXP_SALES(t)	(i) Main ba		()	verage	() ···	in bank	. ,	verage		in bank		verage	· · · ·	in bank	()	verage		in bank	. ,	verage
	dy/dx Std.	dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	-	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.	dy/dx	Std. dev.
BANKINFO (t-1)	0.001 0	0.017			0.026	0.031			-0.004	0.021			0.013	0.023			-0.045	0.049		
BANKINFO_AVR (t-1)			-0.018	0.020			0.023	0.039			-0.029	0.024			-0.011	0.027			-0.105	0.053 **
BANKBR (t-1)	0.000 0	0.000			0.000	0.000			0.000	0.000			0.000	0.000			0.000	0.000		
BANKBR_AVR (t-1)			0.000	0.000			0.000	0.000			0.000	0.000			0.000	0.000			0.000	0.000
FFORINV (t-1)	-0.084 0	0.014 ***	-0.084	0.014 ***	0.046	0.049	0.044	0.049	-0.098	0.014 ***	-0.097	0.014 ***	-0.101	0.014 ***	-0.100	0.014 ***	0.022	0.076	0.026	0.076
JBIC (t-1)	0.022 0	0.013 *	0.023	0.013 *	0.013	0.017	0.012	0.018	0.013	0.020	0.015	0.020	0.013	0.020	0.014	0.020	N.A.	N.A.	N.A.	N.A.
B_SIZE (t-1)	-0.001 0	0.001	-0.001	0.001	-0.001	0.002	-0.003	0.002 *	-0.001	0.001	-0.001	0.001	-0.001	0.001	-0.001	0.001	0.002	0.002	0.001	0.002
B_CAP (t-1)	-0.039 0	0.076	-0.059	0.073	-0.125	0.168	-0.190	0.162	-0.003	0.083	-0.004	0.081	-0.007	0.094	-0.010	0.092	0.177	0.179	0.164	0.177
B_LTD (t-1)	0.008 0	0.004 *	0.008	0.004 *	0.007	0.007	0.009	0.007	0.010	0.006 *	0.010	0.006 *	0.011	0.006 *	0.012	0.006 *	0.002	0.023	0.002	0.023
F NEARBYFIRM (t-1)	-3.5E-06 7.3	E-06 -	-3.3E-06	7.3E-06	-3.0E-06	1.3E-05	-2.8E-06	1.3E-05	-2.1E-06	9.9E-06	-2.3E-06	9.9E-06	-8.4E-06	1.1E-05	-8.5E-06	1.1E-05	1.6E-05	3.4E-05	1.3E-05	3.4E-05
F_NEARBYINDEXP (t-1)		0.000	0.000	0.000	-0.001	0.001	-0.001	0.001	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	-0.001	0.002	-0.001	0.002
IMPORTRATIO (t-1)	-0.022 0	0.019	-0.022	0.019	-0.137	0.038 ***	-0.136	0.038 ***	0.003	0.022	0.002	0.022	-0.066	0.025 ***	-0.067	0.025 ***	0.293	0.053 ***	0.292	0.053 ***
FOREIGN (t-1)			3.5E-05	1.2E-05 ***	9.0E-05	2.6E-05 ***	8.8E-05	2.6E-05 ***	1.4E-05	1.3E-05	1.5E-05	1.3E-05	2.5E-05	1.4E-05 °	2.6E-05	1.4E-05 °	-7.0E-05	3.8E-05 °		3.8E-05 *
TFP (t-1)	0.023 0	0.010 **	0.023	0.010 **	0.018	0.022	0.017	0.022	0.018	0.011	0.018	0.011	0.003	0.013	0.003	0.013	0.036	0.027	0.036	0.027
F_CASH (t-1)	0.006 0	0.014	0.007	0.014	-0.001	0.031	0.000	0.031	0.008	0.016	0.009	0.016	-0.007	0.018	-0.007	0.018	0.041	0.039	0.045	0.039
Firm fixed effect	Yes		Y	'es	Y	'es	Y	les		'es	Y	'es		es	Y	les		'es		es
Year effect	Yes		Y	'es	Y	'es	Y	les	Y	'es	Y	es	Y	es	Y	<i>l</i> 'es	Y	es	Y	es
Number of obs.		19,86				5,3					536			11,3					169	
Number of Groups		4,778	8			1,2	22			3,8	816			3,0)64			1,	077	
Obs. per group: Min		1				1	l				1			1					1	
Average		4.2				4.				3				3.					2.9	
Max		10				1					0				0				10	
F-value	163.78			3.76		3.14		3.24		.26		5.4		.01		3.02		3.72		.91
Prob > F	0.0000			0000		0000		0000		000		0000		000		0000		000		000
R-sq: Within	0.2070			2070		2708		2711		761		763		844		1845		722	0.1	
Between	0.0109			106)231)228		0156		0154	0.0			0111		323		313
Overall	0.0482			1478 0290)758)202		0754 0209		9411 0189)409 0197		354		0351)448)505	0.0	463)456
Corr (u_i, Xb)	-0.0280)	-0.0	1290	-0.0	J202	-0.0	0209	-0.0	1109	-0.0	J19/	-0.0	0451	-0.0	0461	-0.0	1303	-0.0	430
F test that all u_i=0:	12.56		10		10	2.04	1.0	2.06	10	.42	10	2.43	10	.01	1.0	2.02	11	.55	11	57
F-value Prob > F	0.0000			000		2.04		2.06		0000		2.45		.01		2.02		.55		.57
F100 > F	0.0000		0.0	000	0.0	000	0.0	000	0.0	000	0.0	000	0.0	000	0.0	0000	0.0	000	0.0	000

Table 7: Estimation Results for EXP_SALES

Note: ***, **, and * indicate significance at the 1, 5, and 10% level, respectively.

To summarize, the results obtained in the empirical analysis suggest that information on foreign markets provided via various channels, especially through lender banks, substantially reduces the fixed costs of exporting. The findings thus highlight that channels of information spillovers other than those examined in the literature so far may be of considerable importance. Moreover, information spillovers from lender banks appear to be particularly important for SMEs, which typically have fewer transaction partners (suppliers and customers) than larger firms and lack the internal resources to gather information on export markets.

5. Concluding Remarks

This study examined the role of information spillovers through lender banks in the context of firms' export decisions. To do so, we used a unique dataset containing information not only on Japanese firms' export activities and the presence of nearby exporting firms, but also on their lender banks' experience in transacting with other exporting firms and lender banks' own overseas activities. The estimation results indicate that information spillovers through the banks positively affect SMEs' decision to start exporting and expand their export destinations. The information spillovers also reduce the likelihood of firms stopping to export. The export-to-sales ratio of exporters, however, is not affected by such information spillovers. These results imply that information on foreign markets provided by lender banks substantially reduces the fixed entry costs of exporting as well as the costs associated with maintaining firms' export status.

The research presented in this study could be expanded in a number of directions. One such direction would be to extend our analysis to examine other important dimensions of firms' international activities such as foreign direct investment. A further potentially interesting extension would be to use the model in this study to analyze how the impact of changes in currency exchange rates interacts with information spillovers through lender banks. A depreciation of the yen can be expected to encourage Japanese firms to start exporting or increase their exports. Therefore, information spillovers through lender banks may have a larger positive impact on firms' export decision during a period when the home currency is weak than when it is strong. We believe all of these extensions would provide further insights to gain a better understanding of firms' export dynamics and the role of lender banks.

This study also provides an important policy implication. As mentioned in the introduction, even though export promotion has been a key policy tool in many countries, our knowledge as to what factors are important in determining whether a firm becomes an exporter still remains very limited. With regard to Japan, many firms, particularly SMEs, do not export even though they are sufficiently profitable and/or actively invest in research and development. Promoting exports by these firms is an urgent policy issue for Japan, given demographic trends and the prolonged sluggishness of domestic demand. This study has highlighted the importance of banks' role as providers of information for potential exporters, especially SMEs, suggesting that the government should proactively involve banks in its export promotion policies.

On the other hand, banks may also be interested in providing more support services for firms trying to expand their business abroad. In fact, small banks in particular face a situation in which their client firms are experiencing declining domestic demand and are therefore concerned that their own business may shrink. Helping such banks to build international service networks and building on the support services provided by banks may allow the government to implement its export promotion policies more effectively. Moreover, since banks have accumulated substantial information on their client firms' business, they may have useful knowledge on what type of firms should receive support from the government and on what type of support is most effective. Of course, government and non-profit organizations already provide various support services for firms' international business and for trading companies. Information provided by such organizations or trading companies is complementary to information collected by banks through lending relationships, and it is important for the government to effectively utilize these various information sources for export promotion policies.

References

- Aitken, H., G. H. Hanson, and A. E. Harrison (1997), 'Spillovers, Foreign Investment, and Export Behavior', *Journal of International Economics* 43 (1-2), pp. 103-132.
- Amiti, M. and D. Weinstein (2011), 'Exports and Financial Shocks', *Quarterly Journal* of Economics, 126 (4), pp.1841-1877.
- Baldwin, R. and P. Krugman (1989), "Persistent Trade Effects of Large Exchange Rate Shocks," *Quarterly Journal of Economics* 104 (4): 635-654.
- Barrios, S., H. Gorg, and E. Strobl (2003), 'Explaining Firms' Export Behaviour: R&D, Spillovers and the Destination Market', Oxford Bulletin of Economics and Statistics 65 (4), pp.475-496.
- Bernard, A. B. and J. B. Jensen (2004), 'Why do Firms Export', *Review of Economics* and Statistics 86 (2), pp.561-569.
- Bernard, A. B., J. Eaton, J. B. Jensen, and S. Kortum (2003), 'Plants and Productivity in International Trade', *American Economic Review* 93 (4), pp.1268-1290.
- Degryse, H., M. Kim and S. Ongena (2009), *Microeconometrics of Banking: Methods, Applications, and Results*, Oxford: Oxford University Press.
- Feenstra, R. C., Z. Li and M. Yu (2013), 'Exports and Credit Constraints under Incomplete Information: Theory and Evidence from China', *Review of Economics and Statistics*, forthcoming.
- Fukao, K., T. Inui, K. Ito, Y, Kim, and T. Yuan (2011), 'An International Comparison

of the TFP Levels and the Productivity Convergence of Japanese, Korean, Taiwanese, and Chinese Listed Firms', *Journal of Chinese and Economic and Business Studies*, 9(2), pp.127-150.

- Good, D., M. Nadiri, and R. Sickles (1997), 'Index Number and Factor Demand Approaches to the Estimation of Productivity', in Pesaran, H. and P. Schmidt, (eds.), *Handbook of Applied Econometrics: Microeconomics*, Oxford: Blackwell, pp.14-80
- Hoshi, T. and A. Kashyap (2001), *Corporate Financing and Governance in Japan: The Road to the Future.* Cambridge: Mass MIT Press.
- Japanese Bankers Association (2011), Ajia Keizai-ken ni totte Nozomashii Kinyu-Hoken Shijo no Arikata: Ginko no Torihikisaki Kigyo no Kaigai Shinshutsu ni okeru Shien Jisseki [The Shape of a Desirable Financial and Capital Market for the Asian Economy: Case Studies on Banks' Services to Support Client Firms' International Activities], March 2011, Japanese Bankers Association.
- Japan Small Business Research Institute (2008), White Paper on Small and Medium Enterprises in Japan 2008: Improvement of Productivity and the Challenge of Community Revitalization, The Small and Medium Enterprise Agency.
- Koenig, P., F. Mayneris and S. Poncet (2010), 'Local Export Spillovers in France', *European Economic Review* 54, pp.622-641.
- Krautheim, S. (2012), 'Heterogeneous Firms, Exporter Networks and the Effect of Distance on International Trade', *Journal of International Economics* 87(1), pp.27-35.
- Khwaja, A. I. and A. Mian (2008), 'Tracing the Impact of Bank Liquidity Shocks: Evidence from an Emerging Market', American Economic Review, 98, pp.1413-1412.
- Manova, K., S. Wei and Z. Zhang (2011), 'Firm Exports and Multinational Activity under Credit Constraints', NBER Working Paper 16905.
- Mayer, T. and G. Ottaviano (2008), 'The Happy Few: The Internationalisation of European Firms', *Intereconomics* 43, pp.135-148.
- Melitz, M. (2003), 'The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity', *Econometrica* 71(6), pp.1695-1725.
- Minetti, R. and S. C. Zhu (2011), 'Credit Constraints and Firms Export: Microeconomic Evidence from Italy', *Journal of International Economics* 83, pp.109-125.
- Miyakawa, D., K, Hosono, T. Uchino, M. Hazawa., A. Ono., H. Uchida and I. Uesugi

(2013), 'Natural Disasters, Financial Shocks, and Firm Export', mimeo.

- Paravisini, D., V. Rappoport, P. Schnabl and D. Wolfenzon (2011), 'Dissecting the Effect of Credit Supply on Trade: Evidence from Matched Credit-Export Data', NBER Working Paper 16975.
- Patrick, H. (1994), 'The Relevance of Japanese Finance and Its Main Bank System', in Aoki, M. and H. Patrick (eds.), *The Japanese Main Bank System: Its Relevance for Developing and Transforming Economies*, Oxford: Oxford University Press, pp.359-360.
- Todo, Y. (2011), 'Quantitative Evaluation of Determinants of Export and FDI: Firm-Level Evidence from Japan', *World Economy* 34 (3), pp.355-381.

Appendix 1: Summary Statistics

(a) All firms

		All firms			Exporters		1	Non-exporter	s
Variable	Obs.	Average	Std. dev.	Obs.	Average	Std. dev.	Obs.	Average	Std. dev.
TFP	77,305	0.009	0.156	22,526	0.050	0.162	54,779	-0.008	0.150
F_CASH	77,305	0.553	0.173	22,526	0.580	0.154	54,779	0.542	0.179
F_NEARBYFIRM	77,305	310.452	534.795	22,526	428.008	608.762	54,779	262.111	493.162
F_NEARBYINDEXP	77,305	2.488	5.611	22,526	4.667	7.196	54,779	1.592	4.515
FOREIGN	77,305	5.774	56.487	22,526	14.565	88.610	54,779	2.160	35.062
IMPORTRATIO	70,680	0.019	0.068	21,529	0.046	0.095	49,151	0.008	0.046
FFORINV	77,305	0.012	0.050	22,526	0.033	0.076	54,779	0.004	0.029
BANKINFO	77,305	0.217	0.074	22,526	0.240	0.071	54,779	0.207	0.074
BANKINFO_AVR	77,305	0.214	0.059	22,526	0.237	0.057	54,779	0.204	0.057
BANKBR	77,305	15.049	21.236	22,526	17.932	22.416	54,779	13.863	20.615
BANKBR_AVR	77,305	12.920	12.280	22,526	15.116	12.589	54,779	12.018	12.035
B_SIZE	77,305	16.428	1.539	22,526	16.718	1.496	54,779	16.308	1.541
B_CAPRATIO	77,305	0.044	0.014	22,526	0.043	0.014	54,779	0.044	0.014
B_LTD	77,305	0.578	0.202	22,526	0.599	0.233	54,779	0.570	0.188
JBIC	77,305	0.002	0.047	22,526	0.005	0.067	54,779	0.001	0.036

(b) Large firms

		All firms			Exporters			Non-exporters	
Variable	Obs.	Average	Std. dev.	Obs.	Average	Std. dev.	Obs.	Average	Std. dev.
TFP	9,778	0.083	0.149	5,876	0.101	0.150	3,902	0.056	0.144
F_CASH	9,778	0.525	0.154	5,876	0.537	0.143	3,902	0.507	0.167
F_NEARBYFIRM	9,778	553.905	650.779	5,876	600.068	663.208	3,902	484.389	625.287
F_NEARBYINDEXP	9,778	4.377	6.902	5,876	5.519	6.879	3,902	2.656	6.574
FOREIGN	9,778	16.823	80.969	5,876	22.328	90.707	3,902	8.533	62.653
IMPORTRATIO	9,136	0.036	0.088	5,663	0.049	0.096	3,473	0.014	0.068
FFORINV	9,778	0.034	0.062	5,876	0.051	0.072	3,902	0.009	0.030
BANKINFO	9,778	0.252	0.079	5,876	0.261	0.078	3,902	0.237	0.079
BANKINFO_AVR	9,778	0.246	0.062	5,876	0.255	0.061	3,902	0.232	0.060
BANKBR	9,778	19.247	22.452	5,876	20.354	22.744	3,902	17.580	21.902
BANKBR_AVR	9,778	16.111	12.458	5,876	16.890	12.527	3,902	14.938	12.262
B_SIZE	9,778	16.900	1.448	5,876	17.011	1.410	3,902	16.733	1.489
B_CAPRATIO	9,778	0.042	0.013	5,876	0.041	0.013	3,902	0.043	0.013
B_LTD	9,778	0.640	0.321	5,876	0.645	0.318	3,902	0.632	0.324
JBIC	9,778	0.007	0.085	5,876	0.010	0.099	3,902	0.004	0.060

(c) SMEs

		All firms			Exporters		Ν	Non-exporte	rs
Variable	Obs.	Average	Std. dev.	Obs.	Average	Std. dev.	Obs.	Average	Std. dev.
TFP	67,527	-0.002	0.154	16,650	0.032	0.162	50,877	-0.013	0.150
F_CASH	67,527	0.557	0.175	16,650	0.595	0.155	50,877	0.544	0.179
F_NEARBYFIRM	67,527	275.200	506.236	16,650	367.286	576.229	50,877	245.064	477.287
F_NEARBYINDEX	67,527	2.214	5.343	16,650	4.366	7.281	50,877	1.510	4.306
FOREIGN	67,527	4.175	51.801	16,650	11.825	87.697	50,877	1.671	31.926
IMPORTRATIO	61,544	0.017	0.064	15,866	0.044	0.095	45,678	0.008	0.044
FFORINV	67,527	0.009	0.047	16,650	0.026	0.077	50,877	0.003	0.029
BANKINFO	67,527	0.211	0.072	16,650	0.233	0.066	50,877	0.204	0.073
BANKINFO_AVR	67,527	0.209	0.056	16,650	0.231	0.053	50,877	0.202	0.056
BANKBR	67,527	14.441	20.985	16,650	17.077	22.236	50,877	13.578	20.485
BANKBR_AVR	67,527	12.458	12.185	16,650	14.490	12.552	50,877	11.794	11.988
B_SIZE	67,527	16.359	1.540	16,650	16.614	1.511	50,877	16.276	1.540
B_CAPRATIO	67,527	0.044	0.014	16,650	0.043	0.014	50,877	0.044	0.014
B_LTD	67,527	0.570	0.177	16,650	0.583	0.191	50,877	0.565	0.172
JBIC	67,527	0.001	0.039	16,650	0.003	0.052	50,877	0.001	0.033

(d) Medium-sized firms

		All firms			Exporters			Non-exporters	
Variable	Obs.	Average	Std. dev.	Obs.	Average	Std. dev.	Obs.	Average	Std. dev.
TFP	45,298	0.012	0.154	12,959	0.043	0.160	32,339	0.000	0.150
F_CASH	45,298	0.556	0.174	12,959	0.593	0.151	32,339	0.541	0.180
F_NEARBYFIRM	45,298	291.272	518.487	12,959	377.732	581.793	32,339	256.626	486.549
F_NEARBYINDEX	45,298	2.382	5.699	12,959	4.434	7.554	32,339	1.559	4.502
FOREIGN	45,298	5.417	58.654	12,959	13.465	92.927	32,339	2.192	36.364
IMPORTRATIO	41,357	0.019	0.067	12,354	0.045	0.094	29,003	0.008	0.047
FFORINV	45,298	0.011	0.053	12,959	0.028	0.082	32,339	0.004	0.034
BANKINFO	45,298	0.216	0.073	12,959	0.234	0.067	32,339	0.208	0.073
BANKINFO_AVR	45,298	0.213	0.056	12,959	0.232	0.054	32,339	0.206	0.056
BANKBR	45,298	15.144	21.307	12,959	17.451	22.400	32,339	14.219	20.781
BANKBR_AVR	45,298	13.046	12.211	12,959	14.779	12.497	32,339	12.351	12.024
B_SIZE	45,298	16.438	1.530	12,959	16.651	1.503	32,339	16.352	1.532
B_CAPRATIO	45,298	0.044	0.014	12,959	0.043	0.014	32,339	0.044	0.014
B_LTD	45,298	0.577	0.197	12,959	0.587	0.204	32,339	0.573	0.194
JBIC	45,298	0.002	0.043	12,959	0.003	0.058	32,339	0.001	0.036

(e) Small firms

		All firms			Exporters		Ν	Non-exporte	rs
Variable	Obs.	Average	Std. dev.	Obs.	Average	Std. dev.	Obs.	Average	Std. dev.
TFP	22,229	-0.031	0.150	3,691	-0.008	0.162	18,538	-0.036	0.147
F_CASH	22,229	0.559	0.176	3,691	0.602	0.167	18,538	0.550	0.177
F_NEARBYFIRM	22,229	242.448	478.649	3,691	330.610	554.773	18,538	224.894	460.002
F_NEARBYINDEX	22,229	1.873	4.514	3,691	4.127	6.223	18,538	1.424	3.939
FOREIGN	22,229	1.644	33.633	3,691	6.068	65.821	18,538	0.763	22.120
IMPORTRATIO	20,187	0.013	0.057	3,512	0.043	0.100	16,675	0.007	0.039
FFORINV	22,229	0.005	0.030	3,691	0.020	0.057	18,538	0.002	0.020
BANKINFO	22,229	0.203	0.071	3,691	0.228	0.063	18,538	0.198	0.071
BANKINFO_AVR	22,229	0.200	0.056	3,691	0.225	0.051	18,538	0.195	0.055
BANKBR	22,229	13.009	20.238	3,691	15.764	21.603	18,538	12.461	19.910
BANKBR_AVR	22,229	11.262	12.044	3,691	13.473	12.689	18,538	10.821	11.862
B_SIZE	22,229	16.199	1.548	3,691	16.485	1.532	18,538	16.142	1.545
B_CAPRATIO	22,229	0.045	0.014	3,691	0.044	0.014	18,538	0.045	0.014
B_LTD	22,229	0.554	0.125	3,691	0.571	0.137	18,538	0.551	0.123
JBIC	22,229	0.001	0.027	3,691	0.000	0.016	18,538	0.001	0.028

Appendix 2: BANKINFO Variable

The table shows the distribution of *BANKINFO* for the top 76 banks as of the end of FY2000 in our dataset. Banks are sorted in descending order in terms of *BANKINFO*. *NUM_CLIENT* is the number of total client firms of each bank.

Ranking	NUM_CLIENT	BANKINFO	Ranking	NUM_CLIENT	BANKINFO
1	126	0.44	39	780	0.21
2	76	0.41	40	3,033	0.20
3	56	0.38	41	54	0.20
4	62	0.34	42	69	0.20
5	3,347	0.31	43	499	0.20
6	1,670	0.30	44	508	0.20
7	7,035	0.30	45	3,312	0.19
8	1,232	0.30	46	493	0.19
9	58	0.29	47	208	0.18
10	453	0.29	48	4,544	0.18
11	2,110	0.28	49	83	0.18
12	378	0.28	50	504	0.18
13	107	0.27	51	100	0.18
14	616	0.27	52	553	0.18
15	828	0.27	53	73	0.18
16	9,582	0.26	54	377	0.18
17	1,109	0.26	55	97	0.18
18	7,492	0.26	56	263	0.17
19	1,196	0.26	57	975	0.17
20	55	0.25	58	476	0.17
21	402	0.25	59	279	0.17
22	1,044	0.25	60	143	0.17
23	4,705	0.24	61	54	0.17
24	206	0.24	62	186	0.17
25	167	0.24	63	642	0.17
26	71	0.24	64	716	0.16
27	3,234	0.24	65	147	0.16
28	1,384	0.24	66	295	0.16
29	416	0.24	67	136	0.16
30	143	0.23	68	94	0.16
31	561	0.22	69	208	0.16
32	185	0.22	70	1,400	0.16
33	224	0.21	71	57	0.16
34	571	0.21	72	541	0.16
35	260	0.21	73	552	0.15
36	128	0.21	74	145	0.15
37	171	0.21	75	179	0.15
38	627	0.21	76	317	0.15

ENDNOTES

* This research was conducted as part of research projects for the Economic Research Institute for ASEAN and East Asia (ERIA), the Research Institute of Economy, Trade and Industry (RIETI), and the Economic and Social Research Institute (ESRI). This work was supported by JSPS KAKENHI

Grant Numbers 23243050 and 23243044. The opinions expressed and arguments employed in this paper are the sole responsibility of the authors and do not necessarily reflect those of ERIA, RIETI, or any institution to which the authors are affiliated.

[†] Professor, College of Economics, Nihon University, 1-3-2, Misaki-cho, Chiyoda-ku, Tokyo, 101-8360, JAPAN. E-mail: inui.tomohiko@nihon-u.ac.jp.

[‡] Professor, School of Economics, Senshu University. 2-1-1 Higashi-mita, Tama-ku, Kawasaki, Kanagawa, 214-8580, JAPAN. E-mail: keiko-i@isc.senshu-u.ac.jp.

** Research Associate, Weatherhead Center for International Affairs, Harvard University. 61 Kirkland Street, Cambridge, MA 02138, U.S.A. E-mail: damiyak@gmail.com.

¹ Other strands in the literature examine the relationship between firms' export status and their innovative capacity, the price and/or quality of their products, various country characteristics, and institutional factors such as free trade agreements, economic diplomacy, and so on.

² That financial institutions likely play an important role in determining client firms' export activities has recently been highlighted in studies by Amiti and Weinstein (2011), Paravisini *et al.* (2011), Feenstra *et al.* (2013), and Miyakawa *et al.* (2013).

³ The survey asks for the amount as well as the destination or origin of exports and imports broken down into seven regions (Asia, Middle East, Europe, North America, Latin America, Africa, and Oceania). Unfortunately, more detailed information on the destination of exports and origin of imports is not available.

⁴ Although the survey also asks non-manufacturing firms for information on exports and imports, they are required to provide the amount of trade in goods only. The survey started to ask about international transactions in services in the 2010 survey.

⁵ In this study, we calculate firms' TFP level following Fukao *et al.* (2011), using the multilateral TFP index method developed by Good *et al.* (1997). Specifically, the TFP level of firm *i* in industry *j* in year *t*, *TFP*_{*i*,*j*,*t*} is defined in comparison with the TFP level of a hypothetical representative firm in the benchmark year t_0 in industry *j*. The benchmark year t_0 is set to the year 2000 in this study.

⁶ The first nearby-firm variable, *F_NEARBYFIRM*, represents the number of firms located in the same city for each firm. The second nearby-firm variable, *F_NEARBYINDEXP*, represents the number of exporting firms belonging to the same industry and located in the same city for each firm. ⁷ For an overview of the origins of the main bank system, see, e.g., Hoshi and Kashyap (2001).

⁸ Like Baldwin and Krugman (1989), we assume that once firms have started exporting they need to pay some fixed costs to continue exporting. For example, firms still have to invest in marketing, reputation, distribution, and so on, to remain in export markets. Baldwin and Krugman (1989) call these costs "maintenance cost."

No.	Author(s)	Title	Year
2014-16	Tomoko INUI, Keiko ITO, and Daisuke MIYAKAWA	Japanese Small and Medium-Sized Enterprises' Export Decisions: The Role of Overseas Market Information	July 2014
2014-15	Han PHOUMIN and Fukunari KIMURA	Trade-offRelationshipbetweenEnergyIntensity-thus energy demand- and Income Level:Empirical Evidence and Policy Implications forASEAN and East Asia Countries	June 2014
2014-14	Cassey LEE	The Exporting and Productivity Nexus: Does Firm Size Matter?	May 2014
2014-13	Yifan ZHANG	Productivity Evolution of Chinese large and Small Firms in the Era of Globalisation	May 2014
2014-12	Valéria SMEETS, Sharon TRAIBERMAN, Frederic WARZYNSKI	Offshoring and the Shortening of the Quality Ladder:Evidence from Danish Apparel	May 2014
2014-11	Inkyo CHEONG	Korea's Policy Package for Enhancing its FTA Utilization and Implications for Korea's Policy	May 2014
2014-10	Sothea OUM, Dionisius NARJOKO, and Charles HARVIE	Constraints, Determinants of SME Innovation, and the Role of Government Support	May 2014
2014-09	Christopher PARSONS and Pierre-Louis Vézina	Migrant Networks and Trade: The Vietnamese Boat People as a Natural Experiment	May 2014
2014-08	Kazunobu HAYAKAWA and Toshiyuki MATSUURA	Dynamic Tow-way Relationship between Exporting and Importing: Evidence from Japan	May 2014
2014-07	DOAN Thi Thanh Ha and Kozo KIYOTA	Firm-level Evidence on Productivity Differentials and Turnover in Vietnamese Manufacturing	Apr 2014
2014-06	Larry QIU and Miaojie YU	Multiproduct Firms, Export Product Scope, and Trade Liberalization: The Role of Managerial Efficiency	Apr 2014
2014-05	Han PHOUMIN and Shigeru KIMURA	Analysis on Price Elasticity of Energy Demand in East Asia: Empirical Evidence and Policy Implications for ASEAN and East Asia	Apr 2014

ERIA Discussion Paper Series

No.	Author(s)	Title	Year
2014-04	Youngho CHANG and	Non-renewable Resources in Asian Economies:	Feb
2014-04	Yanfei LI	Perspectives of Availability, Applicability, Acceptability, and Affordability	2014
2014-03	Yasuyuki SAWADA and Fauziah ZEN	Disaster Management in ASEAN	Jan 2014
2014-02	Cassey LEE	Competition Law Enforcement in Malaysia	Jan 2014
2014-01	Rizal SUKMA	ASEAN Beyond 2015: The Imperatives for Further Institutional Changes	Jan 2014
2013-38	Toshihiro OKUBO, Fukunari KIMURA, Nozomu TESHIMA	Asian Fragmentation in the Global Financial Crisis	Dec 2013
2013-37	Xunpeng SHI and Cecilya MALIK	Assessment of ASEAN Energy Cooperation within the ASEAN Economic Community	Dec 2013
2013-36	Tereso S. TULLAO, Jr. And Christopher James CABUAY	Eduction and Human Capital Development to Strengthen R&D Capacity in the ASEAN	Dec 2013
2013-35	Paul A. RASCHKY	Estimating the Effects of West Sumatra Public Asset Insurance Program on Short-Term Recovery after the September 2009 Earthquake	Dec 2013
2013-34	Nipon POAPONSAKORN and Pitsom MEETHOM	Impact of the 2011 Floods, and Food Management in Thailand	Nov 2013
2013-33	Mitsuyo ANDO	Development and Resructuring of Regional Production/Distribution Networks in East Asia	Nov 2013
2013-32	Mitsuyo ANDO and Fukunari KIMURA	Evolution of Machinery Production Networks: Linkage of North America with East Asia?	Nov 2013
2013-31	Mitsuyo ANDO and Fukunari KIMURA	What are the Opportunities and Challenges for ASEAN?	Nov 2013
2013-30	Simon PEETMAN	Standards Harmonisation in ASEAN: Progress, Challenges and Moving Beyond 2015	Nov 2013
2013-29	Jonathan KOH and Andrea Feldman MOWERMAN	Towards a Truly Seamless Single Windows and Trade Facilitation Regime in ASEAN Beyond 2015	Nov 2013
2013-28	Rajah RASIAH	Stimulating Innovation in ASEAN Institutional Support, R&D Activity and Intelletual Property Rights	Nov 2013

No.	Author(s)	Title	Year			
2012.27	Maria Monica	Financial Integration Challenges in ASEAN	Nov			
2013-27	WIHARDJA	beyond 2015	2013			
		Who Disseminates Technology to Whom, How,	N			
2013-26	Tomohiro MACHIKITA and Yasushi UEKI	and Why: Evidence from Buyer-Seller Business	Nov 2013			
		Networks				
2013-25	Fukunari KIMURA	Reconstructing the Concept of "Single Market a	Oct			
2013-25		Production Base" for ASEAN beyond 2015	2013			
	Olivier CADOT Ernawati MUNADI	Streamlining NTMs in ASEANI	0-4			
2013-24	Lili Yan ING	Streamlining NTMs in ASEAN:	Oct			
		The Way Forward	2013			
	Charles HARVIE,	Small and Medium Enterprises' Access to	Oct			
2013-23	Dionisius NARJOKO,	Finance: Evidence from Selected Asian	2013			
	Sothea OUM	Economies	2013			
2013-22	Alan Khee-Jin TAN	Toward a Single Aviation Market in ASEAN:	Oct			
2013-22	Aldii Kilee-Jili IAN	Regulatory Reform and Industry Challenges	2013			
	Hisanobu SHISHIDO,	Moving MPAC Forward: Strengthening				
2013-21	Shintaro SUGIYAMA,	Public-Private Partnership, Improving Project	Oct			
2013-21	Fauziah ZEN	Portfolio and in Search of Practical Financing	2013			
		Schemes				
	Barry DESKER, Mely	Thought/Issues Paper on ASEAN Food Security:	Oct			
2013-20	CABALLERO-ANTH	Towards a more Comprehensive Framework				
	ONY, Paul TENG	Towards a more comprehensive Pranework				
	Toshihiro KUDO,	Making Myanmar the Star Growth Performer in	Sep			
2013-19	Satoru KUMAGAI, So	ASEAN in the Next Decade: A Proposal of Five	2013			
	UMEZAKI	Growth Strategies	2010			
		Managing Economic Shocks and	Sep			
2013-18	Ruperto MAJUCA	Macroeconomic Coordination in an Integrated	2013			
		Region: ASEAN Beyond 2015	2010			
2013-17	Cassy LEE and Yoshifumi	Competition Policy Challenges of Single Market	Sep			
2010 17	FUKUNAGA	and Production Base	2013			
2013-16	Simon TAY	Growing an ASEAN Voice? : A Common	Sep			
2013 10		Platform in Global and Regional Governance	2013			
2013-15	Danilo C. ISRAEL and	Impacts of Natural Disasters on Agriculture, Food	Aug			
	Roehlano M. BRIONES	Security, and Natural Resources and Environment in	2013			

No.	Author(s)	Title	Year
		the Philippines	
2013-14	Allen Yu-Hung LAI and Seck L. TAN	Impact of Disasters and Disaster Risk Management in Singapore: A Case Study of Singapore's Experience in Fighting the SARS Epidemic	Aug 2013
2013-13	Brent LAYTON	Impact of Natural Disasters on Production Networks and Urbanization in New Zealand	Aug 2013
2013-12	Mitsuyo ANDO	Impact of Recent Crises and Disasters on Regional Production/Distribution Networks and Trade in Japan	Aug 2013
2013-11	Le Dang TRUNG	Economic and Welfare Impacts of Disasters in East Asia and Policy Responses: The Case of Vietnam	Aug 2013
2013-10	Sann VATHANA, Sothea OUM, Ponhrith KAN, Colas CHERVIER	Impact of Disasters and Role of Social Protection in Natural Disaster Risk Management in Cambodia	Aug 2013
2013-09	Sommarat CHANTARAT, Krirk PANNANGPETCH, Nattapong PUTTANAPONG, Preesan RAKWATIN, and Thanasin TANOMPONGPHANDH	Index-Based Risk Financing and Development of Natural Disaster Insurance Programs in Developing Asian Countries	Aug 2013
2013-08	Ikumo ISONO and Satoru KUMAGAI	Long-run Economic Impacts of Thai Flooding: Geographical Simulation Analysis	July 2013
2013-07	Yoshifumi FUKUNAGA and Hikaru ISHIDO	Assessing the Progress of Services Liberalization in the ASEAN-China Free Trade Area (ACFTA)	May 2013
2013-06	Ken ITAKURA, Yoshifumi FUKUNAGA, and Ikumo ISONO	A CGE Study of Economic Impact of Accession of Hong Kong to ASEAN-China Free Trade Agreement	May 2013
2013-05	Misa OKABE and Shujiro URATA	The Impact of AFTA on Intra-AFTA Trade	May 2013
2013-04	Kohei SHIINO	How Far Will Hong Kong's Accession to ACFTA will Impact on Trade in Goods?	May 2013
2013-03	Cassey LEE and Yoshifumi FUKUNAGA	ASEAN Regional Cooperation on Competition Policy	Apr 2013
2013-02	Yoshifumi FUKUNAGA	Taking ASEAN+1 FTAs towards the RCEP:	Jan

No.	Author(s)	Title	Year
	and Ikumo ISONO	A Mapping Study	2013
2013-01	Ken ITAKURA	Impact of Liberalization and Improved Connectivity and Facilitation in ASEAN for the ASEAN Economic Community	Jan 2013
2012-17	Sun XUEGONG, Guo	Market Entry Barriers for FDI and Private Investors:	Aug
2012-17	LIYAN, Zeng ZHENG	Lessons from China's Electricity Market	2012
2012-16	Yanrui WU	Electricity Market Integration: Global Trends and Implications for the EAS Region	Aug 2012
2012-15	Youngho CHANG, Yanfei LI	Power Generation and Cross-border Grid Planning for the Integrated ASEAN Electricity Market: A Dynamic Linear Programming Model	Aug 2012
2012-14	Yanrui WU, Xunpeng SHI	Economic Development, Energy Market Integration and Energy Demand: Implications for East Asia	Aug 2012
2012-13	Joshua AIZENMAN, Minsoo LEE, and Donghyun PARK	The Relationship between Structural Change and Inequality: A Conceptual Overview with Special Reference to Developing Asia	July 2012
2012-12	Hyun-Hoon LEE, Minsoo LEE, and Donghyun PARK	Growth Policy and Inequality in Developing Asia: Lessons from Korea	July 2012
2012-11	Cassey LEE	Knowledge Flows, Organization and Innovation: Firm-Level Evidence from Malaysia	June 2012
2012-10	Jacques MAIRESSE, Pierre MOHNEN, Yayun ZHAO, and Feng ZHEN	Globalization, Innovation and Productivity in Manufacturing Firms: A Study of Four Sectors of China	June 2012
2012-09	Ari KUNCORO	Globalization and Innovation in Indonesia: Evidence from Micro-Data on Medium and Large Manufacturing Establishments	June 2012
2012-08	Alfons PALANGKARAYA	The Link between Innovation and Export: Evidence from Australia's Small and Medium Enterprises	June 2012
2012.07	Chin Hee HAHN and	Direction of Causality in Innovation-Exporting Linkage:	June
2012-07	Chang-Gyun PARK	Evidence on Korean Manufacturing	2012
2012-06	Keiko ITO	Source of Learning-by-Exporting Effects: Does Exporting Promote Innovation?	June 2012
2012-05	Rafaelita M. ALDABA	Trade Reforms, Competition, and Innovation in the	June

No.	Author(s)	Title	Year
		Philippines	2012
2012-04	Toshiyuki MATSUURA and Kazunobu HAYAKAWA	The Role of Trade Costs in FDI Strategy of Heterogeneous Firms: Evidence from Japanese Firm-level Data	June 2012
2012-03	Kazunobu HAYAKAWA, Fukunari KIMURA, and Hyun-Hoon LEE	How Does Country Risk Matter for Foreign Direct Investment?	Feb 2012
2012-02	Ikumo ISONO, Satoru KUMAGAI, Fukunari KIMURA	Agglomeration and Dispersion in China and ASEAN: A Geographical Simulation Analysis	Jan 2012
2012-01	Mitsuyo ANDO and Fukunari KIMURA	How Did the Japanese Exports Respond to Two Crises in the International Production Network?: The Global Financial Crisis and the East Japan Earthquake	Jan 2012
2011-10	Tomohiro MACHIKITA and Yasushi UEKI	Interactive Learning-driven Innovation in Upstream-Downstream Relations: Evidence from Mutual Exchanges of Engineers in Developing Economies	Dec 2011
2011-09	Joseph D. ALBA, Wai-Mun CHIA, and Donghyun PARK	Foreign Output Shocks and Monetary Policy Regimes in Small Open Economies: A DSGE Evaluation of East Asia	Dec 2011
2011-08	Tomohiro MACHIKITA and Yasushi UEKI	Impacts of Incoming Knowledge on Product Innovation: Econometric Case Studies of Technology Transfer of Auto-related Industries in Developing Economies	Nov 2011
2011-07	Yanrui WU	Gas Market Integration: Global Trends and Implications for the EAS Region	Nov 2011
2011-06	Philip Andrews-SPEED	Energy Market Integration in East Asia: A Regional Public Goods Approach	Nov 2011
2011-05	Yu SHENG, Xunpeng SHI	Energy Market Integration and Economic Convergence: Implications for East Asia	Oct 2011
2011-04	Sang-Hyop LEE, Andrew MASON, and Donghyun PARK	Why Does Population Aging Matter So Much for Asia? Population Aging, Economic Security and Economic Growth in Asia	Aug 2011
2011-03	Xunpeng SHI,	Harmonizing Biodiesel Fuel Standards in East Asia:	May

No.	Author(s)	Title	Year
	Shinichi GOTO	Current Status, Challenges and the Way Forward	2011
2011-02	Hikari ISHIDO	Liberalization of Trade in Services under ASEAN+n : A Mapping Exercise	May 2011
2011-01	Kuo-I CHANG, Kazunobu HAYAKAWA Toshiyuki MATSUURA	Location Choice of Multinational Enterprises in China: Comparison between Japan and Taiwan	Mar 2011
2010-11	Charles HARVIE, Dionisius NARJOKO, Sothea OUM	Firm Characteristic Determinants of SME Participation in Production Networks	Oct 2010
2010-10	Mitsuyo ANDO	Machinery Trade in East Asia, and the Global Financial Crisis	Oct 2010
2010-09	Fukunari KIMURA Ayako OBASHI	International Production Networks in Machinery Industries: Structure and Its Evolution	Sep 2010
2010-08	Tomohiro MACHIKITA, Shoichi MIYAHARA, Masatsugu TSUJI, and Yasushi UEKI	Detecting Effective Knowledge Sources in Product Innovation: Evidence from Local Firms and MNCs/JVs in Southeast Asia	Aug 2010
2010-07	Tomohiro MACHIKITA, Masatsugu TSUJI, and Yasushi UEKI	How ICTs Raise Manufacturing Performance: Firm-level Evidence in Southeast Asia	Aug 2010
2010-06	Xunpeng SHI	Carbon Footprint Labeling Activities in the East Asia Summit Region: Spillover Effects to Less Developed Countries	July 2010
2010-05	Kazunobu HAYAKAWA, Fukunari KIMURA, and Tomohiro MACHIKITA	Firm-level Analysis of Globalization: A Survey of the Eight Literatures	Mar 2010
2010-04	Tomohiro MACHIKITA and Yasushi UEKI	The Impacts of Face-to-face and Frequent Interactions on Innovation: Upstream-Downstream Relations	Feb 2010
2010-03	Tomohiro MACHIKITA and Yasushi UEKI	Innovation in Linked and Non-linked Firms: Effects of Variety of Linkages in East Asia	Feb 2010

No.	Author(s)	Title	Year
2010-02	Tomohiro MACHIKITA and Yasushi UEKI	Search-theoretic Approach to Securing New Suppliers: Impacts of Geographic Proximity for Importer and Non-importer	Feb 2010
2010-01	Tomohiro MACHIKITA and Yasushi UEKI	Spatial Architecture of the Production Networks in Southeast Asia: Empirical Evidence from Firm-level Data	Feb 2010
2009-23	Dionisius NARJOKO	Foreign Presence Spillovers and Firms' Export Response: Evidence from the Indonesian Manufacturing	Nov 2009
2009-22	Kazunobu HAYAKAWA, Daisuke HIRATSUKA, Kohei SHIINO, and Seiya SUKEGAWA	Who Uses Free Trade Agreements?	Nov 2009
2009-21	Ayako OBASHI	Resiliency of Production Networks in Asia: Evidence from the Asian Crisis	Oct 2009
2009-20	Mitsuyo ANDO and Fukunari KIMURA	Fragmentation in East Asia: Further Evidence	Oct 2009
2009-19	Xunpeng SHI	The Prospects for Coal: Global Experience and Implications for Energy Policy	Sept 2009
2009-18	Sothea OUM	Income Distribution and Poverty in a CGE Framework: A Proposed Methodology	Jun 2009
2009-17	Erlinda M. MEDALLA and Jenny BALBOA	ASEAN Rules of Origin: Lessons and Recommendations for the Best Practice	Jun 2009
2009-16	Masami ISHIDA	Special Economic Zones and Economic Corridors	Jun 2009
2009-15	Toshihiro KUDO	Border Area Development in the GMS: Turning the Periphery into the Center of Growth	May 2009
2009-14	Claire HOLLWEG and Marn-Heong WONG	Measuring Regulatory Restrictions in Logistics Services	Apr 2009
2009-13	Loreli C. De DIOS	Business View on Trade Facilitation	Apr 2009
2009-12	Patricia SOURDIN and Richard POMFRET	Monitoring Trade Costs in Southeast Asia	Apr 2009
2009-11	Philippa DEE and Huong DINH	Barriers to Trade in Health and Financial Services in ASEAN	Apr 2009

No.	Author(s)	Title	Year
2009-10	Sayuri SHIRAI	The Impact of the US Subprime Mortgage Crisis on the World and East Asia: Through Analyses of Cross-border Capital Movements	Apr 2009
2009-09	Mitsuyo ANDO and Akie IRIYAMA	International Production Networks and Export/Import Responsiveness to Exchange Rates: The Case of Japanese Manufacturing Firms	Mar 2009
2009-08	Archanun KOHPAIBOON	Vertical and Horizontal FDI Technology Spillovers:Evidence from Thai Manufacturing	Mar 2009
2009-07	Kazunobu HAYAKAWA, Fukunari KIMURA, and Toshiyuki MATSUURA	Gains from Fragmentation at the Firm Level: Evidence from Japanese Multinationals in East Asia	Mar 2009
2009-06	Dionisius A. NARJOKO	Plant Entry in a More LiberalisedIndustrialisationProcess: An Experience of Indonesian Manufacturing during the 1990s	Mar 2009
2009-05	Kazunobu HAYAKAWA, Fukunari KIMURA, and Tomohiro MACHIKITA	Firm-level Analysis of Globalization: A Survey	Mar 2009
2009-04	Chin Hee HAHN and Chang-Gyun PARK	Learning-by-exporting in Korean Manufacturing: A Plant-level Analysis	Mar 2009
2009-03	Ayako OBASHI	Stability of Production Networks in East Asia: Duration and Survival of Trade	Mar 2009
2009-02	Fukunari KIMURA	The Spatial Structure of Production/Distribution Networks and Its Implication for Technology Transfers and Spillovers	Mar 2009
2009-01	Fukunari KIMURA and Ayako OBASHI	International Production Networks: Comparison between China and ASEAN	Jan 2009
2008-03	Kazunobu HAYAKAWA and Fukunari KIMURA	The Effect of Exchange Rate Volatility on International Trade in East Asia	Dec 2008
2008-02	Satoru KUMAGAI, Toshitaka GOKAN, Ikumo ISONO, and Souknilanh KEOLA	Predicting Long-Term Effects of Infrastructure Development Projects in Continental South East Asia: IDE Geographical Simulation Model	Dec 2008

No.	Author(s)	Title	Year
2008-01	Kazunobu HAYAKAWA,	Firm-level Analysis of Globalization: A Survey	Dec 2008
	Fukunari KIMURA, and		
	Tomohiro MACHIKITA		